NOWALSKY, BRONSTON & GOTHARD, L.L.P. ATTORNEYS AT LAW

LEON I. NOWALSKY BENJAMIN W. BRONSTON EDWARD P. GOTHARD 3500 N. CAUSEWAY BOULEVARD SUITE 1442 METAIRIE, LOUISIANA 70002 TELEPHONE: (504) 832-1984 FACSIMILE: (504) 831-0892

MONICA R. BORNE JEFFREY T. GREENBERG Of Counsel

4460

March 16, 1998

Florida Public Service Commission Capital Circle Office Center 2540 Shumard Oak Blvd. Tallahassee, Florida 32399-0850

980370-TI

Re:

Notification by IntraTel Group, Ltd. and North American Telephone Network, L.L.C. of Agreement and Plan of Merger

Dear Sir

On behalf of IntraTel Group, Ltd. ("IntraTel"), and North American Telephone Network, L.L.C. ("NATN"), this letter is to advise the Commission of an Agreement and Plan of Merger (the "Agreement") which the companies propose to consummate. It is our understanding, based upon review of the applicable statutes and regulations, that this transaction does not require prior Commission approval. Accordingly, absent written notice to the contrary within thirty (30) days, the parties will proceed to consummate the transaction in a timely fashion.

IntraTel and NATN propose an Agreement and Plan of Merger whereby NATN will become a wholly owned subsidiary of IntraTel, and the present equity owners of NATN ("Members") will receive consideration consisting of cash plus additional contingent cash consideration depending upon the performance of NATN during the year immediately following consummation of the proposed transaction. In addition, NATN will change its name from "North American Telephone Network, L.L.C." to "North American Telephone Network, Inc."

IntraTel is a publicly held Delaware corporation with principal offices located at 28050 U.S. Highway 19 North, Suite 202, Clearwater, Florida 34621. IntraTel is a holding company which does not directly offer long distance telecommunications services, but rather owns and operates other wholly owned subsidiaries which offer such services pursuant to the FCC's Competitive Carrier policies.

NATN is a privately held Georgia limited liability company with principal offices located at 4151 Ashford Dunwoody Road, Suite 550, Atlanta, Georgia, 30319-1462. NATN is a non-dominant carrier that resells domestic interstate and international long distance service purchased from various facilities based carriers pursuant to the FCC's Competitive Carrier policies. NATN is authorized by the FCC to offer domestic interstate and international services in all fifty (50) states and the District of Columbia as a non-dominant carrier. NATN currently originates interstate and international traffic in forty eight (48) states, and provides intrastate service, pursuant to certification, registration or tariff requirements, or on an unregulated basis, in forty eight (48) states NATN is a certificated carrier in the State of Florida.

Applicants propose a transaction which will accomplish the following:

- The present record and beneficial holders of 100% of the issued and outstanding equity membership interests of NATN shall transfer and deliver to IntraTel all of their right, title and interest in and to said interests;
- IntraTel shall issue and deliver to the present NATN Members cash and contingent consideration (collectively, "Purchase Consideration") as provided for and specifically set forth in the Agreement, Exhibit "A";
- The Purchase Consideration shall be allocated to the NATN record and beneficial Members on the basis of the percentage ownership of each Member.

In Florida, NATN provides intrastate telecommunications services pursuant to certification under Docket Number T960313-TI, Order Number PSC-96-0700-FOF-TI, effective date June 14, 1996

- NATN will become, by virtue of the Agreement, a wholly owned subsidiary of IntraTel; and
- NATN shall continue to operate as a regulated entity pursuant to its present certifications, registrations, 'ariff requirements and rate structures, or on an unregulated basis, as provided by and pursuant to applicable law.

Critical to the proposed Agreement is the need to ensure the continuation of high quality service to all customers currently served by NATN. The proposed transaction will serve the public interest for several reasons. First, it will enable the companies to operate with a more streamlined and efficient level of service for all involved customers by creating a larger parent operation. The transaction will enhance the operating efficiencies, including market efficiencies, of both Applicants. Second, it will increase the appeal to present and potential customers of NATN as can provide telecommunications services to its customers in a more efficient manner. I mally, it may result in cost savings because of discounts on quantity ordering of materials and solices for both NATN and IntraTel's other subsidiaries.

Current financial information for IntraTel is attached hereto as Exhibit "B". Current financial information for NATN is attached hereto as Exhibit "C".

The parties are forwarding this letter to the Commission for informational purposes, to be included in the appropriate files. Absent receipt of written notification to the contrary within thirty (30) days, we will proceed on our understanding that no approval or other formal action is required by the Commission prior to consummation of the proposed transaction.

Exhibit "B" is the Form 10-Q submitted to the SEC by IntraTel for the period ended September 30, 1997

Exhibit "C" consists of copies of NATN's unaudited financial statements for the years 1995, 1996 and 1997

Enclosed are the original and thirteen (13) copies of this letter. Please return one (1) of the copies file-stamped in the envelope provided. If you need any further information or have any questions regarding the matters discussed herein, please 40 not hesitate to contact the undersigned.

Thank you for your assistance in this matter.

Respectfully submitted,

Benjamin W. Bronston

EllenAnn G. Sands

Nowalsky, Bronston & Gothard, L.L.P.

3500 North Causeway Boulevard

Suite 1442

Metairie, Louisiana 70002

(504) 832-1984

Counsel for IntraTel Group, Ltd. and

North American Telephone Network, L.L.C.

Charles D. Ganz.

Holland & Knight, LLP

One Atlantic Center, Suite 2000

1201 West Peachtree Street, N.E.

Atlanta, Georgia 30369-3400

Telephone: (404) 898-81 '2

Counsel for North Aperican Telephone

Network, L.L.C.

STATE OF LOUISIANA

PARISH OF JEFFERSON

VERIFICATION

I, Hans Kasper, am the President of North American Telephone Network, L.L.C., and am authorized to make this verification on its behalf. The statements made in the foregoing Application are true of my own knowledge, except as to those matters which are therein stated on information and belief, and as to those matters I believe them to be true.

By:

: Hans Kasper

Title: F

President

Sworn to and subscribed before me, Notary Public, in and for the State and Parish named above, this 13th day of March, 1998.

Notary Public

My commission expires at death

STATE OF LOUISIANA

PARISH OF JEFFERSON

VERIFICATION

I, Murray L. Swanson, am the President and Chief Executive Officer of IntraTel Group, Ltd., and am authorized to make this verification on its behalf. The statements made in the foregoing Application are true of my own knowledge, except as to those matters which are therein stated on information and belief, and as to those matters I believe them to be true.

Name: Murray L. Swanson

Title: President and Chief Executive Officer

Sworn to and subscribed before me, Notary Public, in and for the State and Parish named above, this 13th day of March, 1998.

Notary Public

My commission expires at death

EXHIBIT A

AGREEMENT AND PLAN OF MERGER

THIS IS AN AGREEMENT AND PLAN OF MERGER (this "Agreement"), dated as of February 27, 1998, by and among INTRATEL GROUP, LTD., a Delaware corporation ("IntraTel"), INTRATEL ACQUISITION/NATN CORPORATION, a Delaware corporation ("SubCorp"), NORTH AMERICAN TELEPHONE NETWORK, L.L.C., a Georgia limited liability company ("NATN"), and each of the members of NATN, who are listed on the signature pages of this Agreement (collectively, the "Members" and individually, a "Member"). NATIONAL SERVICE DIRECT, INC., a Georgia corporation ("NSDI"), has joined as a party to this Agreement for the limited purpose of agreeing to be bound by Section 7.9 of this Agreement

RECITALS:

IntraTel and NATN desire to effect a business combination of SubCorp and NATN pursuant to which NATN will merge with and into SubCorp (the "Merger"), and the holders of membership interests in NATN ("NATN Membership Interests"), will receive cash in exchange for such interests on the terms and conditions described herein. In furtherance thereof, (i) the managers and the Members of NATN have approved the Merger; and (ii) the Board of Directors of IntraTel and the Board of Directors and sole shareholder of SubCorp have approved the Merger.

NOW, THEREFORE, in consideration of the mutual representations, warranties, covenants and agreements contained herein, the parties hereto agree as follows:

ARTICLE 1

THE BUSINESS COMBINATION

- 1.1 The Merger. At the Effective Time (as defined in Section 1.2 hereof), NATN shall be merged with and into SubCorp in accordance with the provisions of this Agreement, the applicable provisions of the Delaware General Corporation Law (the "DGCL"), and the applicable provisions of the Georgia Limited Liability Company Act (the "GLLCA"), and the separate existence of NATN shall thereupon cease, and SubCorp, as the surviving corporation in the Merger (the "Surviving Corporation"), shall continue its corporate existence under the laws of the State of Delaware.
- 1.2 Effective Time of the Merger. As soon as practicable following fulfillment of the conditions set forth in Article 8, and provided that this Agreement has not been terminated pursuant to Article 10, NATN and SubCorp (collectively the "Constituent Corporations") shall cause appropriate Articles of Merger to be duly prepared, executed and filed with the Secretary of State of Delaware, in accordance with the provisions of the DGCL, and with the Secretary of State of Georgia, in accordance with the provisions of the GLLCA. The Merger shall become effective at the time specified in the Articles of Merger (the "Effective Time").

1.3 Effect of the Merger. At and after the Effective Time, the Merger shall have all of the effects provided by applicable law.

ARTICLE 2

CONVERSION OF MEMBERSHIP INTERESTS; MERGER CONSIDERATION; ADDITIONAL ACTIONS

- 2.1 <u>Conversion of NATN Membership Interests</u>. At the Effective Time and as a result of the Merger:
- (a) <u>NATN Membership Interests</u>. Each NATN Membership Interest shall, automatically and without further act of any of the Constituent Corporations, IntraTel or the holder of any such NATN Membership Interest, be converted into and shall thereafter represent only the right to receive the Merger Consideration as set forth in Section 2.2 and shall not thereafter represent any other rights.
- (b) <u>SubCorp Common Stock</u>. Each issued and outstanding share of common stock of SubCorp shall remain unchanged.
- 2.2 <u>Merger Consideration</u>. The "Merger Consideration" shall be \$40,000,000.00 (plus or minus any amount due based on the formula set forth in <u>Exhibit 2.3</u> hereto), payable in cash as follows:
- (a) Payments at Closing. At the Closing, SubCorp shall pay to the Members by wire transfer of immediately available funds an amount equal to \$20,000,000.00 (the "Closing Payment"). The Closing Payment shall be paid to each Member in proportion to such Member's ownership of NATN Membership Interests as set forth on Schedule 2.2(a) hereto.
- (b) <u>Contingent Payment</u>. As of each "Determination Date" (as defined herein) and as soon as its amount is finally determined pursuant to Section 2.4 below, the Surviving Corporation shall pay to the Members an additional payment of up to \$20,000,000.00 (plus or minus any amount due based on the formula set forth in Exhibit 2.3 hereto) in cash (the "Contingent Payment"), calculated pursuant to Section 2.3 below.
 - 2.3 Calculation of Contingent Payment.
 - (a) As used in this Agreement:
- (i) "Annualized EBITDA" shall mean, at any Determination Date (I) the total EBITDA earned by the Surviving Corporation (including, if applicable,

NATN prior to the Effective Time) during the three-month period immediately preceding such Determination Date, multiplied by (II) four (4)

- (ii) "Annualized Revenue" shall mean, at any Determination Date (I) the total revenue earned by the Surviving Corporation (including, if applicable, NATN prior to the Effective Time) during the three-month period immediately preceding such Determination Date, multiplied by (II) four (4)
- (iii) "calendar quarter" means any three-month period ending on any March 31, June 30, September 30 or December 31.
- (iv) "Determination Date" shall mean and include (A) the last day of each calendar quarter during the Earn-Out Period, (B) the last day of the month during which the Earn-Out Period expires, (C) in the event the Revenue Goal is achieved prior to the expiration of the Earn-Out Period, the last day of the month during which the Surviving Corporation achieves such Revenue Goal, and (D) if the Management Agreement to be entered into pursuant to Section 8.1(e) of this Agreement is terminated prior to the expiration of the Earn-Out Period, the last day of the month during which such termination occurs.
- (v) "Earn-Out Period" means the twelve-month period immediately following the Effective Time.
- (vi) "EBITDA" means total Earnings Before Interest, Taxes, Depreciation and Amortization, calculated in accordance with GAAP.
- (vii) "EBITDA Earn-Out Amount" shall mean, at any Determination Date, the result obtained by (I) adding Annualized EBITDA to Past Year's EBITDA, and (II) by dividing the sum of such amounts by two i.e., -

Annualized EBITDA + Past Year's EBITDA 2

- (viii) "EBITDA Earn-Out Margin" shall mean, at any Determination Date, the result obtained by dividing the EBITDA Earn-Out Amount by the Revenue Earn-Out Amount.
- (ix) "GAAP" means generally accepted accounting principles as interpreted by the Securities and Exchange Commission consistently applied.
- (x) "Majority of the Members" shall be determined by reference to a majority percentage of the NATN Membership Interests as set forth on Schedule 2.2(a) hereto, and may be constituted by as few as one Member holding more than 50% of the NATN Membership Interests.

- (xi) "Management Period" means the period from the Effective Time until the date that any and all Contingent Payment due hereunder has been finally determined and fully paid in accordance with the terms of this Agreement.
- (xii) "Past Year's EBITDA" shall mean, at any Determination Date, the total EBITDA earned by Surviving Corporation (including, if applicable, NATN prior to the Effective Time) during the 12-month period immediately preceding such Determination Date.
- (xiii) "Past Year's Revenue" shall mean, at any Determination Date, the total revenue earned by Surviving Corporation (including, if applicable, NATN prior to the Effective Time) during the 12-month period immediately preceding such Determination Date.
- (xiv) "Revenue Earn-Out Amount" shall mean, at any Determination Date, the result obtained by (I) adding Annualized Revenue to Past Year's Revenue, and (II) by dividing the sum of such amounts by two <u>i.e.</u>, -

Annualized Revenue + Past Year's Revenue

- (xv) "Revenue Goal" means a Revenue Earn-Out Amount equal to or greater than \$70,000,000.
- (b) (i) The Contingent Payment due to the Members as of any Determination Date (other than a Determination Date occurring upon the termination of the Management Agreement pursuant to subclause (D) of Section 2.3(a)(iv) shall be the amount determined pursuant to the table set forth on Exhibit 2.3(a) hereto less (1) the cumulative amount of all Contingent Payments previously paid to the Members pursuant to this Agreement and (II) in the case of any Contingent Payment due to the Members at the end of the Earn-Out Period, all Customer Acquisition Expense Advances (as defined in Section 7.10(a), which have not been repaid by the Surviving Corporation to IntraTel as of the end of the Earn-Out Period; provided, however, that for the purposes of determining the amount of such Customer Acquisition Expense Advances, if any, to be so deducted from the Contingent Payment amount determined pursuant to this clause (II), Customer Acquisition Expense Advances shall not include any sum that IntraTel shall have advanced to the Surviving Corporation to replace sums that would have been available to the Surviving Corporation pursuant to NATN's existing credit facilities or similar credit facilities.
- (ii) The amount of the Contingent Payment due to the Members as of any Determination Date occurring upon the termination of the Management Agreement pursuant to subclause (D) of Section 2.3(a)(iv) shall be the amount determined pursuant to the table set forth on Exhibit 2.3(b) hereto less deducting (I)

the cumulative amount of all Contingent Payments previously paid to the Members pursuant to this Agreement and (II) all Customer Acquisition Expense Advances which have not been repaid by the Surviving Corporation to IntraTel as of such termination of the Management Agreement; provided, however, that for the purposes of determining the amount of such Customer Acquisition Expense Advances, if any, to be so deducted from the Contingent Payment amount determined pursuant to this clause (II), Customer Acquisition Expense Advances shall not include any sum that IntraTel shall have advanced to the Surviving Corporation to replace sums that would have been available to the Surviving Corporation pursuant to NATN's existing credit facilities or similar credit facilities.

- Determination of Contingent Payment. Within 15 days after each Determination Date, the Members shall deliver to IntraTel the Members' calculation of the Contingent Payment due to the Members for the relevant period. If IntraTel disagrees with the Members' calculation of the amount of the Contingent Payment due to the Members, IntraTel shall notify the Members of such disagreement within 15 days after the receipt of such calculation, specifying in detail the reasons for such disagreement. If, within 20 days after the receipt by the Members of IntraTel's notice of disagreement, IntraTel and a Majority of the Members are unable to reach agreement on the amount of the Contingent Payment due to the Members, IntraTel and a Majority of the Members shall select an independent public accounting firm not used by either IntraTel or the Surviving Corporation and such accounting firm shall determine within 90 days after the end of the relevant Determination Date the amount of the Contingent Payment due to the Members pursuant to this Agreement. The Surviving Corporation, on the one hand, and the Members, on the other hand, shall share equally the cost of such independent accounting firm. If IntraTel and a Majority of the Members cannot jointly agree upon such an independent public accounting firm, then either IntraTel or a Majority of the Members may apply to the Atlanta, Georgia office of the American Arbitration Association for the appointment of such an independent accounting firm.
- 2.5 Security for Contingent Payment: Payment of Contingent Payment. The Contingent Payment shall be secured by a letter of credit issued by a financial institution reasonably acceptable to the Members, in form, scope and substance reasonably satisfactory to the Members, which shall, as of the Effective Time, be in the face amount of \$5,000,000.00, with the Members as the beneficiaries thereof. Within 15 days after receiving the Members' calculation of any Contingent Payment that would be due following any Determination Date described in subclause (A) of Section 2.3(a)(iv), IntraTel shall either (i) cause to be delivered to the Members a supplemental letter of credit in such an amount that, when added to existing letters of credit, shall equal the total sum that would be due to all of the Members if the Contingent Payment were to be made to the Members as of the end of such calendar quarter, or (ii) pay to the Members in cash the amount of such supplemental letter of credit. Notwithstanding the foregoing, upon the earlier of any Determination Date described in subclauses (B), (C) or (D) of Section 2.3 (a)(iv) the Contingent Payment earned

through such Determination Date will be due in full in cash within 15 days of IntraTel receiving the Members' calculation of the Contingent Payment earned through such Determination Date. Any Contingent Payment due hereunder shall be paid by wire transfer of immediately available funds to each Member in proportion to such Member's ownership of NATN Membership Interests as set forth on Schedule 2.2(a) hereto.

2.6 Further Assurances. If at any time after the Effective Time any further assignments or assurances are necessary or desirable to vest or to perfect or confirm of record in the Surviving Corporation the title to any property or rights of either of the Constituent Corporations, or otherwise to carry out the provisions of this Agreement, the officers and directors of the Surviving Corporation are hereby authorized and empowered on behalf of the respective Constituent Corporations, in the name of and on behalf of the appropriate Constituent Corporation, to execute and deliver any and all things necessary or proper to vest or to perfect or confirm title to such property or rights in the Surviving Corporation, and otherwise to carry out the purposes and provisions of this Agreement.

ARTICLE 3

CERTIFICATE OF INCORPORATION, BYLAWS, BOARD DIRECTORS AND OFFICERS

- 3.1 <u>Certificate of Incorporation: Bylaws</u>. The Certificate of Incorporation and Bylaws of SubCorp as in effect immediately prior to the Effective Time shall be the Certificate of Incorporation and Bylaws of the Surviving Corporation.
- 3.2 <u>Directors and Officers</u>. Except as provided in Section 8.3(b) with respect to the election of James K. Nohle, Jr., the directors and officers of SubCorp immediately prior to the Effective Time shall be the directors and officers of the Surviving Corporation, each to hold office until his or her successor is elected and qualified in accordance with the DGCL and the Certificate of Incorporation and Bylaws of the Surviving Corporation.

ARTICLE 4

REPRESENTATIONS AND WARRANTIES OF NATN

NATN represents and warrants to, and agrees with, IntraTel and SubCorp as follows:

4.1 Organization and Qualification, etc. NATN is a limited liability company duly organized, validly existing and in good standing under the laws of the State of

Georgia, has the corporate power and authority to own all of its properties and assets and to carry on its business as it is now being conducted, and is duly qualified to do husiness and is in good standing in the respective jurisdictions shown on Schedule 4.1, which, except as set forth on Schedule 4.1, are the only jurisdictions in which NATN owns any property or has any place of business or where such qualification is required and where the failure so to qualify would have a material adverse effect upon the financial condition of NATN. The copies of NATN's Articles of Organization and Operating Agreement, as amended to date, which have been delivered to IntraTel, are complete and correct, and such instruments, as so amended, are in full force and effect at the date hereof. The minute books, membership records and other corporate record books of NATN are complete, accurate and up to date in all material respects (and were complete, accurate and up to date in all material respects when made available to IntraTel and its representatives for review). NATN has at all times since its formation qualified as a limited liability company and its Members have always conducted the business of NATN through NATN.

- 4.2 <u>Authorization, Etc.</u> NATN has the corporate power and authority to execute and deliver this Agreement and to consummate the transactions contemplated on its part hereby. The execution and delivery by NATN of this Agreement and the consummation by NATN of the transactions contemplated on its part hereby have been duly authorized by all necessary corporate action on the part of NATN, including without limitation, the requisite vote of the Members. This Agreement has been duly executed and delivered by NATN and is a valid and binding agreement of NATN, enforceable against NATN in accordance with its terms, subject to (a) bankruptcy, insolvency and other laws of similar import, (b) principles of equity, and (c) applicable public policy.
- 4.3 NATN Membership Interests. The Members are the sole members of the Company and own all the NATN Membership Interests. NATN has no commitments to issue or sell any membership interests or any securities or obligations convertible into or exchangeable for, or giving any person any right to subscribe for or acquire from NATN, any membership interests, and no securities or obligations evidencing any such rights are outstanding.
- 4.4 Non-Contravention. Except as set forth in Schedule 4.4, the execution and delivery of this Agreement by NATN do not, and the consummation of the transactions contemplated hereby will not: (a) violate any provision of the Articles of Organization or Operating Agreement of NATN, (b) violate any law, governmental rule or regulation or order binding upon NATN or constitute, in any material respect, a breach of or constitute a default under, or result in the acceleration of or entitle any party to accelerate any obligation under, or result in the creation or imposition of any lien, security interest or other encumbrance upon any of the property of NATN pursuant to any provision of, any mortgage, lease, contract or other agreement to which NATN is a party or by which NATN or its assets is or may be bound.

- Governmental and Other Consents. Except as set forth in Schedule 4.5, and except for (i) the filings with the Federal Trade Commission ("FTC") and the Department of Justice ("DOJ") as required by the Hart-Scott-Rodino Antitrust Improvements Act of 1976, as amended (the "HSR Act"), (ii) the filing of Articles of Merger with the Secretary of State of Delaware and the Secretary of State of Georgia. and (iii) the required filings with and approvals by the Federal Communications Commission ("FCC"), (a) no consent, authorization, order or approval of, or filing or registration with any governmental agency or authority is required to be made or obtained by NATN for or in connection with the execution and delivery of this Agreement by NATN and the concummation by NATN of the transactions contemplated hereby; and (b) no consents or other actions are required to be obtained or taken by NATN to permit IntraTel or SubCorn (as the Surviving Corporation in the Merger) to continue to enjoy, in all material respects, the same rights and benefits as NATN now enjoys under any contracts, agreemente and other instruments that are disclosed on any schedule to this Agreement or that are otherwise material to NATN and the conduct of its business.
- 4.6 <u>No Subsidiaries</u>. NATN does not have any Subsidiaries. As used in this Agreement, the term "Subsidiary" means, with respect to any corporation or other entity ("Parent"), any corporation or other organization, whether incorporated or unincorporated, of which at least a majority of the securities or interests having by the terms thereof ordinary voting power to elect a majority of the board of directors or others performing similar functions with respect to such corporation or other organization is at that time directly or indirectly owned or controlled by such Parent, or by any one or more of its Subsidiaries, or by such Parent and one or more of its Subsidiaries. Except as specified in <u>Schedule 4.6</u>, NATN does not directly or indirectly own any interest in any corporation, partnership, joint venture or other non-corporate form of business organization.
- 4.7 Financial Statements of NATN. Within 30 days after the execution of this Agreement NATN will furnish IntraTel with true and complete copies of the audited financial statements of NATN for the years ended December 31, 1997, 1996 and 1995 (collectively, the "Financial Statements"). The Financial Statements, including, the notes thereto will be (i) prepared in conformity with GAAP and will present fairly in all material respects the financial condition of NATN as of their respective dates and the results of operations for the periods then ended, (ii) reported upon by BDO Seidman, independent certified public accountants, whose report letter will be included therewith.
- 4.8 <u>Absence of Certain Changes or Events</u>. Except for this Agreement and the transactions contemplated hereby or as disclosed on <u>Schedule 4.8</u> hereto, since December 31, 1997:

- (a) Ordinary Course. NATN has operated its business in the ordinary and usual course.
- (b) No Damage, etc. There has not been any damage, destruction or other casualty loss with respect to property owned by NATN, whether or not covered by insurance, or any strike, work stoppage or slowdown or other labor trouble involving NATN, which, in any of such cases or in the aggregate, materially and adversely affects the financial condition of NATN.
- (c) No Material Adverse Change. There has not been any material adverse change in the financial condition, results of operations or prospects of NATN.
- (d) No Issuances, etc. There has not been any issuance by NATN of any membership interests, or options, calls or commitments relating to membership interests, or any securities or obligations convertible into or exchangeable for, or giving any person any right to acquire from any of them, any membership interests, or any redemption, purchase or other acquisition by NATN of any such membership interests.
- (e) <u>Material Contracts</u>. NATN has not entered into, amended or terminated any Material Contract (as hereinafter defined).
- (f) <u>Certain Payments</u>. NATN has not made any payment to, or entered into any transaction with, any Affiliate of NATN or to or with the Members of NATN other than (i) compensation, expense reimbursement and similar payments in the ordinary course of business consistent with past practices, (ii) distributions necessary to enable Members to pay their respective <u>pro rata</u> shares of taxes on income earned by NATN, and (iii) distributions not in excess of \$1,000,000 to Members in an amount equal to special bonus credits given to NATN by MCI with respect to 1997 carrier cost payments.
- (g) <u>Property Interests</u>. Except as may have occurred in the ordinary course of its business, NATN has not (i) sold, assigned, leased, transferred, encumbered, granted a security interest in or license with respect to, or disposed of, any of its assets or properties, tangible or intangible, or (ii) waived or released any rights of value, or cancelled, compromised, released or assigned any indebtedness owed to it or any claims held by it.
- (h) <u>Capital Expenditures</u>. Except as may have occurred in the ordinary course of its business, NATN has not made any capital expenditures in excess of \$100,000 individually or \$750,000 in the aggregate.
- (i) Employee Compensation. There has not been any adoption of, or increase in the payments to or benefits under, any profit sharing, bonua, deferred

compensation, savings, insurance, pension, retirement, or other employee benefit plan for or with any employees of NATN.

- Governmental Authorization and Compliance with Laws. Except as set forth in Schedule 4.9: (a) to NATN's knowledge, the business of NATN has been operated in compliance with all applicable laws and governmental rules and regulations. except for violations of such laws, rules or regulations which do not individually or in the aggregate affect materially and adversely the financial condition or business of NATN, (b) to NATN's knowledge, NATN has all permits, certificates, licenses, approvals and other authorizations required in connection with the operation of its business, (c) no notice has been issued and no investigation or review is pending, or to NATN's knowledge, is contemplated or threatened by any governmental entity with respect to any alleged (i) violation by NATN of any such law, rule or regulation, or (ii) failure to have all permits, certificates, licenses, approvals and other authorizations required in connection with the operation of the businesses of NATN. NATN has provided IntraTel with copies of each and every registration, filing or submission with the FCC since January 1, 1996, and each filing or submission with any state public service commission or public utilities commission since January 1, 1996. To NATN's knowledge, such filings or submissions were in material compliance with applicable law when filed and no material deficiencies have been asserted by any such authority with respect to such filings or submissions.
- 4.10 Absence of Undisclosed Liabilities and Obligations. Except with respect to (a) \$307,700 of indebtedness borrowed by NATN from James K. Noble, Jr. and Christopher G. Hodges in December, 1997, (b) obligations incurred in the ordinary course of business to OAN Services, Inc. in connection with that certain Account Purchase Agreement, dated July 17, 1996, as amended, between NATN and OAN Services, Inc., and (c) other obligations incurred in the ordinary course of business which do not exceed \$1,000,000 in the aggregate (consisting of billing charges, acquisition costs, interest expense, long-distance carrier costs, and general accounts payable), NATN has not incurred any debts, liabilities or obligations (whether absolute, contingent, accrued or otherwise) since December 31, 1997. To NATN's knowledge, NATN is not in violation of any instrument, arbitration award, judgment or decree applicable to it or any of its properties.

4.11 Tax Matters. Except as set forth in Schedule 4.11:

(a) Filing of Returns. NATN has filed on a timely basis all returns for Taxes required to have been filed by it and no extensions of time in which to file any such returns are currently in effect. Each such return filed by NATN was true, correct and complete when it was filed and each return for Taxes that will be filed on or before the Effective Time will be true, correct and complete when it is filed.

- (b) <u>Copies of Returns</u>. NATN has delivered to IntraTel true, correct and complete copies of its returns for Taxes for the taxable years ended since January 1, 1995.
- (c) Payment of Taxes. NATN has paid or made provision for the payment of all Taxes, if any, shown as due pursuant to the returns described in paragraph (a) hereof or otherwise required to be paid to any governmental authority or other person and none of such Taxes is delinquent. All tax returns required to be filed by NATN for the period from the date hereof up to and including the Effective Time will be timely filed and all Taxes for such period will be paid by NATN.
- (d) Withholding, etc. All Taxes which NATN is or was required by law to withhold or collect have been duly withheld and collected and have been paid over to the proper governmental authorities in a timely fashion or are held by NATN in a depository bank account for such payment, and all such withholdings and collections and all other payments due in connection therewith are duly set forth on the books of NATN.
- (e) Examinations, etc. NATN has not received notice that any such return has been or is currently being examined by the federal, state or local tax authorities with taxing authority over NATN and its assets and business, and (without limiting the foregoing) all deficiencies asserted as a result of such examinations have been paid or finally settled or are being contested in good faith by appropriate proceedings, and no issue has been raised by such authorities in any such examination which, by application of similar principles, could be expected to result in a proposed deficiency for any other year not so examined or to result in any additional Taxes for years for which returns have not been filed.
- (f) Statute of Limitations. There are no outstanding agreements or waivers extending the statutory period of limitation applicable to any return for Taxes for any period with respect to NATN.
- Taxes upon any of the properties or assets, real or personal, tangible or intangible, of NATN. NATN has not agreed, or is it required, to include in income any adjustment pursuant to Section 481(a) of the Internal Revenue Code of 1986, as amended (the "Code") by reason of a change in accounting method or otherwise. There is no agreement, plan, or arrangement or other contract covering any employee or independent contractor of NATN that could give rise to the payment of any amount that could not be deductible by NATN pursuant to Section 280G of the Code. There is no tax sharing agreement that will require any payment by NATN after the date of this Agreement. For purposes of this Agreement, the term "Taxes" means all taxes, assessments, duties, fees, levies or other governmental charges, including all federal, state, local, foreign or other income, unitary, business, franchise, property, intangible,

withholding, FICA, unemployment compensation, transfer, sales, use, excise and other taxes, assessments, duties, fees, or levies or other governmental charges, of any kind whatsoever, and all deficiency assessments, additions to tax, penalties and interest with respect thereto.

- 4.12 Title to Properties: Absence of Liens and Encumbrances, Etc. Except as set forth in Schedule 4.12, NATN has good title to all of its owned properties and assets, real, personal and mixed, used in its business, free and clear of any liens, charges, pledges, security interests or other encumbrances, except as reflected in the Financial Statements and except for such imperfections of title and encumbrances, if any, as are not substantial in character, amount or extent, and do not materially detract from the value, or interfere with the present use, of the property subject thereto or affected thereby, or otherwise materially impair the business operations of NATN. All leases under which NATN is the lessee of real or personal property are, to the knowledge of NATN, valid and binding on the lessors thereunder in accordance with their respective terms; to the knowledge of NATN, there are no defaults, or conditions under any such lease that, with lapse of time or notice or both, would constitute defaults by NATN or any other party thereto; and consummation of the transactions contemplated hereby will not alter or impair the rights of NATN under any such lease. Except as set forth on Schedule 4.12, NATN has not leased to any other person or entity any of the properties owned by it or subleased to any other person or entity any of the properties leased by it or granted any other person or entity the right to possess any of its properties.
- 4.13 Adequacy: Condition. Except as set forth on Schedule 4.13: (a) the real and personal properties owned or leased by NATN are adequate for the conduct of its business as currently conducted; (b) NATN is not and has not been, to its knowledge, in violation, in any material respect, of any applicable building, zoning, environmental, health, safety or other law, ordinance or regulation in respect of such property, structures or equipment; (c) NATN has not received notice or has knowledge of (i) any pending or contemplated condemnation or eminent domain proceeding affecting such property, (ii) any proposal for increasing the assessed value of such property for state, county, local or other ad valorem taxes, or (iii) any pending or contemplated proceedings or public improvements which could or might result in the levy of any special tax or assessment of a material nature or amount against such property; and (d) there are no outstanding requirements or recommendations by NATN's insurance companies requiring or recommending any repairs or work of a material nature or amount to be done with respect to such property or any basis for such. Except as set forth in Schedule 4.13, consummation of the transactions contemplated by this Agreement will not alter the rights or impair the ability of the Surviving Corporation to use such properties in the conduct of its business as it is now being conduced.

4.14 Material Contracts.

- (a) Schedule 4.14(a) is a list of all employment, incentive compensation, or similar agreements with or for the benefit of any officer, employee or other person which are not subject to cancellation by NATN without penalty or increased cost on not more than 30 days' notice.
- Schedule 4.14(b) lists all Material Contracts to which NATN is a (b) party or by which it is otherwise bound. For purposes of this Agreement, the term "Material Contract" when used with respect to NATN shall mean: (a) all loan and security agreements, guarantees, surety bonds, leases of real property, consulting and employment agreements, and other miscellaneous business agreements material to the operations of NATN, which are to be performed in whole or in part after the Effective Time: (b) any single contract pursuant to which any party thereto is obligated to make payments or provide services after the date of this Agreement aggregating in value more than \$25,000; (c) any contract for the purchase or sale of another business or a significant amount of assets, entered into (i) since January 1, 1997 or (ii) prior to January 1, 1997 if any indemnification, deferred payment or other continuing obligations under such contract are still in effect; and (d) any contract restricting NATN in the nature of a covenant not to compete. Except as disclosed in Schedule 4.14(b), all Material Contracts are in full force and effect and constitute valid and binding obligations of NATN and, to the knowledge of NATN, any other parties thereto. There are no defaults (including any conditions or events that with lapse of time or notice or both would constitute defaults) on the part of NATN in any obligations to be performed by NATN, or to the knowledge of NATN, any other party, pursuant to a Material Contract. Except as disclosed on Schedule 4.14(b), consummation of the transactions contemplated by this Agreement will not require the consent of any other party to any Material Contract. Except as disclosed in Schedule 4.14(b), there are no contracts or options to sell or lease any property or assets, real, personal or mixed, of NATN. NATN has provided IntraTel true and correct copies of all written contracts, plans, etc. required to be listed on Schedule 4.14(b) and complete descriptions of any arrangements required to be identified on Schedule 4.14(b) which are not in writing.
- 4.15 Litigation. Except as disclosed in the Financial Statements or on Schedule 4.15, 'a) there is no claim, action, suit, investigation or proceeding pending or, to the knowledge of NATN, contemplated or threatened against NATN, or against any of its properties, (i) which seeks damages or penalties in excess of \$25,000, (i.) which seeks to prohibit, restrict or delay the transactions contemplated hereby, or (iii) which, in the event of a final determination adverse to NATN, individually or in the aggregate, would materially and adversely affect the financial condition or business of NATN, (b) there is not any judgment, decree, injunction, ruling or order of any court, governmental department, commission, agency or instrumentality, arbitrator or any other person outstanding against NATN involving an amount in excess of \$25,000 or otherwise having any of the effects set forth in clause (a) of this Section, and (c) there is no

charge or complaint filed by any employee of NATN with the Equal Employment Opportunity Commission.

- 4.16 <u>Labor Controversies</u>. NATN is not a party to any collective bargaining agreement, nor has any labor union been recognized as the representative of the employees of NATN, and NATN does not know of any pending, threatened, or contemplated strikes, work stoppage or other labor disputes involving the employees of NATN. NATN has, to the best of its knowledge, complied in all material respects with all legal requirements relating to employment, equal employment opportunity, nondiscrimination, immigration, wages, hours, benefits, collective bargaining, the payment of social security and similar taxes, and occupational safety and health.
- 4.17 Insider Interests. Except as disclosed in the Financial Statements or on Schedule 4.17: (a) no Affiliate, officer or manager, or former officer or former manager of NATN has any outstanding financial obligations to NATN or any agreement. whether formal or informal, with NATN, or any interest in any property, real or personal, tangible or intangible, including without limitation trade names or trademarks used in or pertaining to the business of NATN, except for the normal rights as a member, manager or employee, or has guaranteed any obligations of NATN, and (b) NATN has not incurred any financial obligations to or for the benefit of any Affiliate. officer or manager, or former officer or former manager of NATN except in connection with normal rights as a member or employee. Without in any way limiting the foregoing, except as disclosed on Schedule 4.17, NATN has never requested any manager, officer, employee or agent of NATN to serve as a director, manager, trustee, officer, employee or agent of another domestic or foreign, non-profit or for-profit corporation, limited liability company, partnership, joint venture, trust or other enterprise so as to give rise to any insurance or indemnification obligation by NATN with respect to any liability or expense arising in connection with any such service to any other enterprise.
- 4.18 Intellectual Property. Except as described in Schedule 4.18, NATN owns or possesses adequate rights to use all patents, trademarks, service marks, trade names, copyrights, or applications for the foregoing, and all computer programs, software, databases and documentation used in, relating to, or necessary for the conduct of, the business of NATN (collectively, the "Intellectual Property"). Schedule 4.18 lists: (i) all such Intellectual Property and (ii) all licenses or other agreements (other than licenses of generally available computer programs or software licenses which are immaterial to the business of NATN) pursuant to which NATN has any right to use or enjoy the Intellectual Property that is owned by others (the "Intellectual Property Agreements"). Except as otherwise described on Schedule 4.18, the Intellectual Property owned by NATN is owned free and clear of all claims of others, including present or former emp yees, agents or independent contractors of NATN, and NATN has not received notice that use of such Intellectual Property in its business violates or infringes upon the claimed rights of others. As to the Intellectual Property Agreements: (a) all such

agreements are in full force and effect; (b) neither NATN nor, to the knowledge of NATN, any other party thereto, is in default under any such agreements; (c) NATN is not or will not, by reason of the transactions contemplated by this Agreement, become obligated to make any royalty or similar payment under such agreements; (d) the rights of NATN under such agreements will not be affected by the consummation of the Merger; and (e) NATN has not received notice that the exercise by NATN of its rights under such agreement infringes upon the claimed rights of others and has no knowledge that the rights of NATN with respect to such Intellectual Property are being infringed upon by any other person. In addition, NATN has not received any notice that any of the products or services of NATN infringes upon claimed rights of others. Except as disclosed on Schedule 4.18, NATN has not granted to any person or entity any license or other right to use in any manner any of the Intellectual Property owned by NATN and has not granted any sublicense or right to use any Intellectual Property licensed to NATN under the Intellectual Property Agreements; nor has NATN granted any software licenses or sublicenses that would authorize any person or entity to use any software licensed or sublicensed thereunder for any purpose other than uses limited solely to such person.

4.19 Insurance. Schedule 4.19 summarizes the amount and scope of the insurance currently in force insuring NATN and its properties against loss or liability. All such policies or contracts of insurance are in scope and amount usual and customary for businesses engaged in the business of NATN and are sufficient for compliance with all requirements of law and of all agreements to which NATN is a party. All insurance policies pursuant to which any such insurance is provided are in full force and effect. No notice of cancellation or termination of any insurance policy has been given to NATN and all premiums required to be paid in connection with such insurance policies have been paid in full.

4.20 Benefit Plans.

Employee Benefit Plan is or has been, and NATN has never had an obligation to contribute to, (i) a multiemployer plan within the meaning of Section 3(37) of the Employee Retirement Income Security Act of 1974, as amended ("ERISA") (a "Multiemployer Plan"), (ii) a "multiple employer plan" within the meaning of Section 210 of ERISA Section 413(c) of the Code, or (iii) a "funded welfare plan" within the meaning of Section 419 of the Code. As used in this Agreement, "Employee Benefit Plan" means each pension, retirement, profit-sharing, deferred compensation, stock option, employee stock ownership, severance pay, vacation, bonus or other incentive plan, any other written or unwritten employee program, arrangement, agreement or understanding, whether arrived at through collective bargaining or otherwise, any medical, vision, dental or other health plan, any life insurance plan, or any other employee benefit plan or fringe benefit plan, including, without limitation, any "employee benefit plan," as that term is defined in Section 3(3) of ERISA currently or

previously adopted, maintained by, sponsored in whole or in part by, or contributed to by NATN or any Affiliate thereof for the benefit of employees, retirees, dependents, spouses, directors, managers, independent contractors, or other beneficiaries and under which employees, retirees, dependents, spouses, directors, managers, independent contractors or other beneficiaries are eligible to participate. Except as disclosed on Schedule 4.20, NATN does not currently have, and in the past five years has not had any ERISA Affiliates. As used in this Agreement, "ERISA Affiliate" means (a) any corporation which is a member of a controlled group of corporations, within the meaning of Section 414(b) of the Code, of which NATN is a member, (b) any trade or businesses under common control, within the meaning of Section 414(c) of the Code, of which NATN is a member, and (c) any member of an affiliate service group, within the meaning of Section 414(m) and (o) of the Code, of which NATN or any entity described in clause (a) or (b) is a member.

- (b) With respect to the Employee Benefit Plans listed on Schedule 4.20, NATN has made or will make available to IntraTel for review: (i) correct and complete copies of all such plans and any related trust agreements or other funding arrangements (including insurance contracts), (ii) all determination letters, rulings, opinion letters, information letters, or advisory opinions issued by the United States Internal Revenue Service, the United States Department of Labor, or the Pension Benefit Guaranty Corporation (the "PBGC"), (iii) annual reports or returns, audited or unaudited financial statements, actuarial valuations and reports, and summary annual reports prepared for any Employee Benefit Plan with respect to the most recent three plan years, and (iv) the most recent summary plan descriptions and any material modifications thereto.
- Except as disclosed in Schedule 4.20, all the Employee Benefit Plans and the related trusts subject to ERISA comply with, to the extent applicable, (i) the provisions of ERISA, (ii) all provisions of the Code relating to qualification and tax exemption under Section 401(a) and Section 501(a) of the Code or otherwise applicable to secure intended tax consequences, (iii) all applicable state or federal securities laws, and (iv) all other applicable Laws and collective bargaining agreements and all such Plans and trusts have been administered in compliance with their terms. NATN has not received any notice from any governmental authority questioning or challenging such compliance. All available permits for the Employee Benefit Plans have been obtained, including, but not limited to, timely determination letters on the qualification of the Employee Benefit Plans intended to be qualified under Section 401(e) of the Code and tax exemption of related trusts under Section 501(a) of the Code, and all such permits continue in full force and effect. No event has occurred which will or could give rise to disqualification of any such plan or loss of intended tax consequences under the Code or to any tax under Section 511 of the Code. Except as disclosed on Schedule 4.20, with respect to any Employee Benefit Plan, (i) neither NATN nor any other "disqualified person" or "party in interest," within the meaning of Section 4975 of the

Code and Section 3(14) of ERISA, respectively, has engaged in any "prohibited transaction," within the meaning of Section 4975 of the Code or Section 406 of ERISA, respectively, nor have there been any fiduciary violations under ERISA that could subject NATN (or any officer, director or employee thereof) to any penalty or tax under Section 502(i) of ERISA or Sections 4971 and 4975 of the Code, (ii) no filing, application or other matter is pending with the United States Internal Revenue Service, the PBGC, the United States Department of Labor or any other governmental body, and (iii) there is no action, suit or claim pending (nor, to the knowledge of the NATN, any basis for such a claim), other than routine claims for henefits. Each of the Employee Benefit Plans can be amended, modified or terminated by NATN within a period of thirty (30) days, without payment of any additional compensation or amount or the additional vesting or acceleration of any such benefits, except to the extent that such vesting is required under the Code as a result of the complete or partial termination of any Employee Benefit Plan intended to be qualified within the meaning of Section 401(a) of the Code.

- (d) Except as disclosed in Schedule 4.20, all of the Employee Benefit Plans are in compliance with the applicable reporting and disclosure requirements, and all documents and annual reports or raturns, audited or unaudited financial statements, actuarial valuations, summary annual reports, and summary plan descriptions issued with respect to the Employee Benefit Plans are correct and complete and there have been no changes in the information set forth therein.
- For any Employee Benefit Plan which is an "employee pension benefit plan," as that term is defined in Section 3(2) of ERISA, to the extent applicable, (i) NATN has not filed a notice of intent to terminate any such Plan or adopted any amendment to treat any such Plan as terminated, (ii) the PBGC has not instituted proceedings to terminate any such Plan, and no other event or condition has occurred which might constitute grounds under Section 4042 of ERISA for the termination of, or the appointment of a trustee to administer, any such Plan, (iii) there has been no failure to meet the minimum funding standards of Section 412 of the Code (whether or not waived in accordance with Section 412(d) of the Code) or to make by its due date a required installment under Section 412(m) of the Code with respect to any such Plan, (iv) no accumulated funding deficiency, whether or not waived, exists, and no condition has occurred or exists which by the passage of time would be expected to result in an accumulated funding deficiency, (v) all required premium payments to the PBGC have been paid when due, (vi) no "reportable event," within the meaning of Section 4043 of ERISA (whether or not waived) and no event described in Section 4062(e) or Section 4063 of ERISA has occurred, (vii) no excise taxes are payable under the Code, (viii) no amendment with respect to which security is required under Section 307 of ERISA or Section 401(a)(29) of the Code has been made or is reasonably expected to be made, (ix) there has been no event which could subject NATN to liability under Section 4064 or 4069 or ERISA, and (x) the funding method used in connection with such Plan is acceptable under ERISA and the Code and the actuarial assumptions used in connection with funding such Plan meet the requirement of ERISA and the Code, and the fair

market value of such Plan's assets equals or exceeds the present value of all benefit liabilities (whether vested or not) accrued to date by all present or former participants in such Plan. For this purpose the assumptions prescribed by the PBGC for valuing plan assets or liabilities upon plan termination shall be applied and the term "benefits" shall include the value of all benefits, rights and features protected under Section 411(d)(6) of the Code or its successors and any ancillary benefits (including disability, shutdown, early retirement and welfare benefits) provided under any such employee pension benefit plan.

- (f) NATN has not incurred any liability or taken any action, and does not have any knowledge of any action or event that could cause it to incur any liability (i) under Section 412 of the Code or Title IV of ERISA with respect to any "single-employer plan" within the meaning of Section 4001(a)(15) of ERISA, (ii) on account of a partial or complete withdrawal within the meaning of Sections 4203 and 4205 of ERISA, respectively, with respect to any Multiemployer Plan, or (iii) on account of unpaid contributions to any Multiemployer Plan.
- (g) NATN has not, and does not, maintain an Employee Benefit Plan providing welfare benefits (as defined in ERISA Section 3(1)) to employees after retirement or other separation of service except to the extent required under Part 6 of Title I of ERISA or Code Section 4980B or their successors ("COBRA") and all such Plans that are subject to COBRA have been operated in compliance with the requirements thereof.
- (h) Except as disclosed on Schedule 4.20, neither the execution of this Agreement, nor the consummation of the transactions contemplated hereby, will (i) entitle any current or former employee of NATN to severance pay, unemployment compensation or any payment contingent upon a change in control or ownership of NATN, or (ii) accelerate the time of payment or vesting, or increase the amount, of any compensation due to any such employee or former employee.
- 4.21 Bank Accounts. Schedule 4.21 lists all bank accounts and safe deposit boxes of NATN, specifying the account numbers.
- 4.22 <u>Directors and Officers</u>. <u>Schedule 4.22</u> lists the name of each individual serving as a manager or officer of NATN as of the date hereof.
- 4.23 Accuracy of Schedules. Certificates and Documents. All information concerning NATN contained in this Agreement, in any certificate furnished to IntraTel pursuant hereto and in each schedule attached hereto is both complete (in that, except as otherwise stated therein, it represents all the material information called for by the description of the respective schedule in this Agreement and does not omit to state any material fact necessary to make the statements contained therein not misleading) and accurate in all material respects. All documents furnished to IntraTel pursuant to this

Agreement as being documents described in this Agreement or in any schedule attached hereto are true and correct copies of the documents which they purport to represent. Notwithstanding the foregoing, nothing contained in the Business Plan (as such term is defined in Section 7.10) shall be deemed a representation or warranty of NATN for purposes of this Agreement or otherwise.

ARTICLE 5 REPRESENTATIONS WARRANTIES OF THE MEMBERS

Each of the Members, severally, represents and warrants to, and agrees with, NATN as follows:

- 5.1 Authorization. Etc. Such Member is the lawful owner of his NATN Membership Interest and has, and at the Effective Time will have, good and clear title to such NATN Membership Interest, free of all restrictions on transfer, liens, encumbrances, security interests, equities and claims whatsoever. Such Member has, and at the Effective Time will have, full legal right, power and authority, and all authorization and approval required by law, to enter into this Agreement. This Agreement has been duly executed and delivered by or on behalf of such Member and is a valid and binding agreement of such Member, enforceable against such Member in accordance with its terms.
- 5.2 Non-Contravention. Except as set forth in Schedule 4.4, and except for the governmental and other consents disclosed pursuant to Section 4.5 and set forth on Schedule 4.5, the execution and delivery of this Agreement by such Member does not, and the consummation of the transactions contemplated hereby will not, violate any law, governmental rule or regulation or order binding upon such Member or constitute, in any material respect, a breach of or a default under, or result in the acceleration of or entitle any party to accelerate any obligation under, or result in the creation or imposition of any lien, security interest or other encumbrance upon such Member's NATN Membership Interest pursuant to any provision of, any mortgage, lease, contract or other agreement to which such Member is a party or by which such Member is or may be bound.

ARTICLE 6 REPRESENTATIONS WARRANTIES OF INTRATEL AND SUBCORP

IntraTel and SubCorp, jointly and severally, represent and warrant to, and agree with, NATN as follows:

6.1 Organization and Qualification, Etc. Each of IntraTel and SubCorp is a corporation duly organized, validly existing and in good standing under the laws of the

State of Delaware, and has the corporate power and authority to own its properties and assets and to carry on its business as it is now being conducted.

- 6.2 Authorization. Etc. Each of IntraTel and SubCorp has the corporate power and authority to execute and deliver this Agreement and to consummate the transactions contemplated hereby on the part of IntraTel and SubCorp. The execution and delivery by IntraTel and SubCorp of this Agreement and the consummation by IntraTel and SubCorp of the transactions contemplated on their part hereby have been duly authorized by all necessary corporate action on the part of IntraTel and SubCorp. This Agreement has been duly executed and delivered by IntraTel and SubCorp and is a valid and binding agreement of IntraTel and SubCorp, enforceable against IntraTel and SubCorp in accordance with its terms, subject to (1) bankruptcy, insolvency and other laws of similar import, (b) principles of equity, and (c) applicable public policy.
- IntraTel and SubCorp does not, and consummation of the transactions contemplated hereby will not: (a) violate any provision of the Certificate of Incorporation or Bylaws of IntraTel or SubCorp, (b) violate any law, governmental rule or regulation or order binding upon NATN or constitute, in any material respect, a breach of or constitute a default under, or result in the acceleration of or entitle any party to accelerate any obligation under, or result in the creation or imposition of any lien, security interest or other encumbrance upon any of the property of IntraTel or SubCorp pursuant to any provision of, any mortgage, lease, contract or other agreement to which IntraTel or SubCorp is a party or by which IntraTel, SubCorp or their assets is or may be bound.
- 6.4 Consents, etc. Except for (a) the filings with the FTC and the DOJ as required by the HSR Act, (b) the filing of Articles of Merger with the Secretary of State of Delaware and the Secretary of State of Georgia, and (c) the required filings with and approvals by the FCC and applicable state regulatory agencies, no consent, authorization, order or approval, or filing or registration with, any governmental agency or authority is required to be made or obtained by IntraTel or SubCorp for or in connection with the execution and delivery of this Agreement by IntraTel and SubCorp and the consummation by IntraTel and SubCorp of the transactions contemplated hereby.
- 6.5 Periodic Reports. IntraTel has filed all reports required to have been filed by it pursuant to Section 13 or 15(d) of the Securities Exchange Act of 1934, as amended. Further, the information contained in IntraTel's Annual Report on Form 10-KSB for the year ended December 31, 1996 (the "IntraTel Form 10-KSB Report") and IntraTel's Quarterly Reports on Form 10-QSB for the quarters ended March 31, 1997, June 30, 1997 and September 30, 1997 (the "IntraTel Quarterly Reports") did not contain any untrue statement of a material fact or omit to state a material fact required to be stated therein or necessary to make the statements therein, in light of the circumstances under which they were made, not misleading.

- Subsidiaries as of December 31, 1995 and 1996 and the related consolidated statements of income, shareholders' equity and cash flows for each of the two years ended December 31, 1995 and 1996, including the notes thereto, certified by Coopers & Lybrand, L.L.P., independent certified public accountants, contained in the IntraTel Form 10-KSB Report and the financial statements, including the notes thereto, contained in the IntraTel Quarterly Reports have been prepared in conformity with generally accepted accounting principles consistently applied and present fairly in all material respects the consolidated financial condition of IntraTel and its Subsidiaries as of such respective dates and their consolidated results of operations for the periods then ended, subject, in the case of any interim financial statements, to normal year-end audit adjustments and any other adjustments described therein.
- 6.7 Absence of Certain Changes or Events. Since December 31, 1996, there has not been (a) any damage, destruction or other casualty loss with respect to property owned by IntraTel or any of its Subsidiaries whether or not covered by insurance, or any strike, work stoppage or slowdown or other labor trouble involving IntraTel or any of its Subsidiaries which, in any of such cases or in the aggregate, materially and adversely affected the financial condition of IntraTel and its Subsidiaries considered as a whole, or (b) any material adverse change in the financial condition, results of operations or prospects of IntraTel and its Subsidiaries, considered as a whole.
- 6.8 Absence of Undisclosed Liabilities and Obligations. Except as set forth on the balance sheet as of September 30, 1997 included in the most recent IntraTel Quarterly Report, neither IntraTel nor any of its Subsidiaries had, as of September 30, 1997, any debts, liabilities or obligations, whether accrued, absolute, contingent or otherwise and whether due or to become due (including without limitation any liabilities resulting from failure to comply with any law applicable to the conduct of its business) required by generally accepted accounting principles to be set forth on the balance sheet or disclosed in the notes thereto.
- 6.9 <u>Litigation</u>. (a) There is no claim, action, suit, investigation or proceeding pending or, to the knowledge of IntraTel or SubCorp, contemplated or threatened against IntraTel, SubCorp, or any of their properties, (i) which seeks damages or penalties in excess of \$25,000, (ii) which seeks to prohibit, restrict or delay the transactions contemplated hereby or any of the conditions to consummation of the transactions contemplated hereby, or (iii) which, in the event of a final adverse determination, individually or in the aggregate, would materially and adversely affect the financial condition of IntraTel, and (b) there is not any judgment, decree, injunction, ruling or order of any court, governmental department, commission, agency or instrumentality, arbitrator or any other person outstanding against IntraTel involving an amount in excess of \$25,000 or otherwise having any of the effects set forth in clause (a) of this Section.

6.10 Accuracy of Schedules, Certificates and Documents. All information concerning IntraTel or SubCorp contained in this Agreement, in any certificate furnished by IntraTel or SubCorp to NATN pursuant hereto and in each schedule attached hereto is both complete (in that, except as otherwise stated therein, it represents all the information called for by the description of the respective schedule in this Agreement and does not omit to state any material fact necessary to make the statements contained therein not misleading) and accurate in all material respects. All documents furnished by IntraTel or SubCorp pursuant to this Agreement as being documents described in this Agreement or in any schedule attached hereto are true and correct copies of the documents which they purport to represent.

ARTICLE 7

ADDITIONAL COVENANTS AND AGREEMENTS

- 7.1 Conduct of Business. NATN covenants that during the period from the date hereof to the Effective Time:
- (a) Operations by NATN in the Ordinary Course of Business. Except as otherwise provided in or contemplated by this Agreement, NATN shall conduct its operations according to its ordinary and usual course of business and use reasonable commercial efforts to: (i) preserve intact its business organization; (ii) keep available the services of its officers, employees and agents; and (iii) maintain satisfactory relationships with licensors, suppliers, distributors, customers and others having business relationships with it. Officers of NATN shall confer with representatives of IntraTel to keep it informed with respect to operational matters of a material nature and to report the general status of the on-going operations of the business of NATN.
- (b) <u>Forbestance by NATN</u>. Except as otherwise provided in or contemplated by this Agreement, NATN shall not, without the prior written consent of IntraTel:
 - (i) except for obligations incurred in the ordinary course of business to OAN Services, Inc. in connection with that certain Account Purchase Agreement, dated July 17, 1996, as amended, between NATN and OAN Services, Inc., take any action which would increase the total indebtedness of NATN for borrowed money (including capitalized leases) by more than \$500,000 or, except in the ordinary course of business, incur any debt, liability or obligation;
 - (ii) other than in the ordinary course of business, assume, guarantee, endorse or otherwise become responsible for the obligations of any other individual, corporation or other entity, or make any loans or advances to any indi- idual, corporation or other entity;

- (iii) declare or make any distribution on, redeem, or purchase or otherwise acquire, any additional NATN Membership Interests, other than (A) distributions made to enable the members of NATN to pay taxes on income earned by NATN, and (B) distributions not in excess of \$1,000,000 to the Members in an amount equal to special bonus credits given to NATN by MCI with respect to 1997 carrier cost payments;
- (iv) issue or sell any additional NATN Membership Interests, or any securities, rights or obligations convertible into or exchangeable for any additional NATN Membership Interests;
- (v) mortgage, p!-dge or otherwise encumber any material property or asset other than in connection with increases in indebtedness for money borrowed permitted by clause (i) of this subsection 7.1(b);
- (vi) other than in the ordinary course of business, sell, lease, transfer or dispose of any of its properties or assets, waive or release any rights of value, or cancel, compromise, release or assign any indebtedness owed to it or any claims held by it;
- (vii) other than in the ordinary course of business, make any investment of a capital nature either by purchase of stock or securities, contributions to capital, property transfers or otherwise, or by the purchase of any property or assets of any other individual, corporation or other entity;
- (viii) make any one capital expenditure in an amount exceeding \$100,000 or any series of capital expenditures in an aggregate amount exceeding \$750,000;
- (ix) other than in the ordinary course of business, enter into or terminate any contract, agreement, plan, or lease, or make any change in any of its contracts, agreements, plans, or leases;
- (x) increase in any material amount the compensation or fringe benefits of any of its officers, employees or agents or pay any pension or retirement allowance not required by any existing Employee Benefit Plan to any such officers, employees or agents, or commit itself to or enter into, amend or modify any employment agreement or any of the existing Employee Benefit Plans or any new Employee Benefit Plan with or for the benefit of any officer, employee or other person;
- (xi) permit any insurance policy naming it as a beneficiary or a loss payable payee to be canceled or terminated or any of the coverage thereunder to lapse, unless simultaneously with such termination or cancellation

replacement policies providing substantially the same coverage are in full force and effect:

- (xii) amend its Articles of Organization or Operating Agreement, except for amendments necessary to reflect changes in the ownership of NATN Membership Interests;
 - (xiii) enter into any collective bargaining agreement;
- (xiv) make any election or file any amended tax return reflecting any position that could result in a material adverse tax consequence to NATN;
- (xv) enter into an agreement to do any of the things described in clauses (i) through (xiv).
- 7.2 HSR Act Filings. As promptly as practicable after the date hereof, NATN and IntraTel shall make any necessary filings with the FTC and the DOJ under the HSR Act, with respect to the transactions contemplated by this Agreement. In connection with such filings NATN and IntraTel shall, in cooperation with each other, and as promptly as reasonably practicable from time to time hereafter, make all such further filings and submissions, and take such further action, as may be required in connection therewith. Each party shall furnish the other all information in its possession necessary for compliance by the other with the provisions of this Section 7.2. No party shall withdraw any such filing or submission prior to the termination of this Agreement without the written consent of the other party. All governmental or third party fees or charges required to be paid in connection with such filings shall be paid by IntraTel.
- shall use reasonable commercial efforts to file all applications and other documents with, and to obtain all consents and approvals from, the FCC and each state public utility commission or public service commission having jurisdiction over the transactions contemplated hereby, in each case as promptly as is practicable. NATN shall take all such actions (other than the payment of money not then due and owing or the provision of other consideration) as are reasonably requested by IntraTel to assist IntraTel in completing all such filings and obtaining all such consents and approvals as IntraTel is required to make and obtain. IntraTel shall take all such actions (other than the payment of money not then due and owing or the provision of other consideration) as are reasonably requested by NATN to assist NATN in completing all filings and obtaining all consents and approvals as NATN is required to make and obtain. All governmental or third party fees or charges required to be paid in connection with such filings or actions shall be paid by IntraTel.

Other Consents, Authorizations, Etc. In additional to those filings and actions required by Section 7.3 of this Agreement, each party hereto will use its best efforts to obtain all consents, authorizations, waivers, orders and approvals from any governmental commission, board or other regulatory body, and to make all related filings and registrations which such party is required to obtain or make for the consummation of the Merger, for IntraTel to control the Surviving Corporation following the Merger and for the Surviving Corporation to continue to enjoy, following the Merger, all of the rights and benefits of NATN, or which such party is required to ohtain or make in connection with the performance by such party of this Agreement and the consummation by it of the transactions contemplated hereby. Each party hereto also will use its best efforts to obtain all consents, authorizations, waivers and approvals from any nongovernmental third party which such party hereto is required to obtain or make pursuant to any contract or agreement to which it is a party for the consummation of the Merger, for IntraTel to control the Surviving Corporation following the Merger and for the Surviving Corporation to continue to enjoy, following the Merger, an of the rights and benefits of NATN, or which such party is required to obtain or make pursuant to any contract to which it is a party in connection with the performance by such party of this Agreement and the consummation by it of the transactions contemplated hereby. Each party hereto also will cooperate fully with the other parties in assisting them to obtain such consents, authorizations, waivers, orders and approvals that the other parties are required to obtain or make. All governmental or third party fees or charges required to he paid in connection with the matters referred to in this Section shall be paid by IntraTel.

7.5 <u>Investigation: Confidentiality.</u>

Investigation by IntraTel. IntraTel may, prior to the Effective (a) Time, make or cause to be made such investigation of the business and properties of NATN and its financial and legal condition as IntraTel reasonably deems necessary or advisable to familiarize itself therewith, provided that such investigation shall not interfere with normal business operations of NATN. NATN agrees to permit IntraTel and its authorized representatives to have or cause them to be permitted to have, after the date hereof and until the Effective Time, reasonable access to the premises, books and records, officers and employees of NATN at reasonable hours for such purposes, and the officers of NATN will furnish IntraTel with such financial and operating data and other information with respect to the business and properties of NATN as IntraTel shall from time to time reasonably request. NATN will permit IntraTel and its representatives, including its accounting firm, to review the work papers of the accounting firm of NATN relating to their review or audit of the Financial Statements. No investigations by IntraTel or its representatives shall affect the representations and warranties of NATN, and each such representation and warranty shall, to the extent set forth in Article 11 hereof, survive the Closing.

- Investigation by NATN. NATN may, prior to the Effective Time. make or cause to be made such investigation of the business and properties of IntraTel and its Subsidiaries, Intelicom International Corporation, Infinet Software, Inc. and any other entities proposed to be combined with or acquired by IntraTel substantially concurrently with the consummation of the transactions contemplated by this Agreement, and their financial and legal condition as NATN reasonably deems necessary or advisable to familiarize itself therewith, provided that such investigation shall not interfere with normal business operations of IntraTel and its Subsidiaries. IntraTel agrees to permit NATN and its authorized representatives to have or cause them to be permitted to have, after the date hereof and until the Effective Time, full access to the premises, books and records, officers and employees of IntraTel and its Subsidiaries at reasonable hours for such purposes, and the officers of IntraTel and its Subsidiaries will furnish NATN with such financial and operating data and other information with respect to the business and properties of IntraTel and its Subsidiaries as NATN shall from time to time reasonably request. IntraTel will permit NATN and its representatives, including its accounting firm, to review the work papers of the accounting firm of IntraTel relating to their review or audit of the financial statements as of and for the three most recent years. No investigations by NATN or its representatives shall affect the representations and warranties of IntraTel and SubCorp, and each such representation and warranty shall, to the extent set forth in Article 11 hereof, survive the Closing.
- Confidentiality. Except as otherwise required in public filings which any party makes with regulatory entities, any information which any party provides to the other or to the other's Representatives, whether written or oral, shall be treated as confidential material (the "Confidential Material"); provided, however, that Confidential Material shall not include information that is generally available to the public or becomes generally available to the public other than as a result of a disclosure by the receiving party or its Representatives. For purposes of this Agreement, the term "Representatives" shall mean a party's directors, managers, officers, employees, attorneys, accountants, investment bankers, brokers, bankers and others engaged by such party or intended to be engaged by such party to advise it regarding the Confidential Material or the transactions contemplated hereby or to assist in financing the transactions contemplated hereby and who receive Confidential Material. It is hereby agreed that the Confidential Material will be used by the receiving party and/or its Representatives only for purposes of evaluating and facilitating the transactions contemplated hereby, and that the Confidential Material will be kept confidential by the receiving party and its Representatives; provided, however, that any of such information may be disclosed to the receiving party's Representatives who need to know such information for purposes relating to transactions contemplated hereby (it being understood that such Representatives shall be informed by the receiving party of the confidential nature of such information and shall be directed by the receiving party to treat such information confidentially). The provisions of this subsection (c) shall remain in effect for a period of five years after the date hereof; provided, however, that

nothing herein shall limit IntraTel's rights to disclose any business information about NATN covered hereby following any consummation of the transactions contemplated hereby.

7.6 Expenses. Except as otherwise provided herein, whether or not the Merger is consummated, all costs and expenses (including any brokerage commissions or any finder's fees and including reasonable attorneys' and accountants' fees) incurred in connection with this Agreement and the transactions contemplated hereby and thereby shall be paid by the party incurring such expenses; provided, however, that IntraTel shall pay the first \$50,000 of fees and expenses of counsel for the Members, NATN and NSDI that may be incurred in connection with negotiation, preparation and execution of that certain Letter of Intent, duted October 2, 1997, among IntraTel, the Members, NATN and NSDI, as amended, and this Agreement and with the Closing and the consummation of the transactions contemplated by this Agreement; provided, further, that, at the Closing, the amount of any such fees and expenses paid by IntraTel (up to \$50,000) may be deducted by IntraTel from the Closing Payment payable to the Members pursuant to Section 2.2(a) hereto. In the event the Closing does not occur. neither NATN, NSDI nor any Member shall have any obligation to return to IntraTel any amounts previously paid by IntraTel to NATN for such fees and expenses.

7.7 No Solicitation of Transactions: Extension Fee.

- defined below) on or before the date of the execution of this Agreement, prior to the close of business on June 1, 1998, neither NATN nor any of its managers, officers, agents or other Representatives shall directly or indirectly (i) solicit, initiate or encourage submission of proposals or offers from any person or entity other than IntraTel relating to any acquisition or purchase of all or a material part (50% or more) of the stock or assets of, or any merger, consolidation or business combination with, NATN (an "Acquisition Proposal"), or (ii) participate in any discussions or negotiations regarding, or furnish to any other person (other than IntraTel and its Representatives) any information with respect to or otherwise cooperate in any way or assist, facilitate or encourage any Acquisition Proposal by any other person. If, notwithstanding the foregoing, NATN should receive any Acquisition Proposal or any inquiry regarding any such proposal from a third party, NATN shall promptly inform IntraTel of the terms of such proposal or inquiry.
- (b) The parties hereto acknowledge and agree that the "Standstill Period" (as such term is defined in that certain Letter of Intent, dated as of October 2, 1997, as amended, among the parties hereto) expired on January 31, 1998. As consideration for NATN agreeing to enter into this Agreement and to agree to the June 1, 1998 date set forth in Section 7.7(a) above, IntraTel shall pay to NATN, on or before the date of the execution of this Agreement, a non-refundable extension fee of \$250,000 (the "Extension Fee") in immediately available funds. In the event the Merger is not

27

consummated pursuant to this Agreement on or before June 1, 1998 (unless such failure to consummate has been made impossible by any act or failure to act by NATN or the Members), NATN shall be entitled to retain such \$250,000 Extension Fee.

- 7.8 Publicity. Except as otherwise required by law, NATN and IntraTel shall coordinate with each other with respect to any notices to third parties, press releases or other public announcements or filings with respect to this Agreement and the transactions contemplated hereby. Neither NATN nor IntraTel shall act unilaterally in this regard, except as may be required by law, without the prior approval of the other party, which approval shall not be unreasonably withheld.
- Joint Venture Between the Surviving Corporation and NSDI. Following the Closing and subject to the occurrence thereof, the Surviving Corporation and NSDI each agree to enter into good faith negotiations for the purpose of reaching mutual agreement on or before December 31, 1998 for the formation of a joint venture entity to be owned equally by the Surviving Corporation and NSDI, which entity shall pursue and establish telemarketing service centers in international markets (other than Canada). The agreement establishing such joint venture shall generally provide that (a) NSDI shall construct such service centers and establish training programs for the joint venture, (b) the Surviving Corporation shall be responsible for the costs of such construction and training, (c) NSDI shall notify the Surviving Corporation of any international market (other than Canada) that NSDI desires to serve and shall give the Surviving Corporation a 30-day option to have such market served by the joint venture rather than NSDI, (d) in the event that the Surviving Corporation determines that such international market should not be served by the joint venture (or the Surviving Corporation fails to respond within such 30-day period), NSDI shall be permitted to serve such international market on its own, (e) all profits of the joint venture shall be split equally between NSDI and the Surviving Corporation, with the calculation of such profits subject to audit rights to be mutually agreed upon in such joint venture agreement, and (f) at the end of three years after the formation of the joint venture, the Surviving Corporation shall have the option to purchase NSDI's 50% interest in the joint venture at a price determined pursuant to terms and conditions which shall be specified in the joint venture agreement; and, in the event the Surviving Corporation does not exercise such option, NSDI shall have the option to purchase the Surviving Corporation's 50% interest in the joint venture at a price determined pursuant to terms and conditions which shall be specified in the joint venture agreement. The foregoing provisions of this Section 7.9 are intended to represent only non-binding understandings of IntraTel, SubCorp and NSDI. In the event that NSDI and the Surviving Corporation are unable to agree upon the terms of such joint venture agreement by December 31, 1998, the obligation to establish such joint venture shall terminate and no party hereto shall have any further rights or obligations pursuant to this Section.

- 7.10 Certain Obligations of IntraTel and the Surviving Corporation following the Effective Time. Except as may be otherwise provided in the Management Agreement to be entered into pursuant to Section 8.1(e) of this Agreement, during the Management Period, IntraTel and the Surviving Corporation shall (i) preserve the separate corporate existence and business organization of the Surviving Corporation, (ii) permit the Surviving Corporation to be managed by the Members (or an entity owned or controlled by the Members) as provided in the Management Agreement to be entered into pursuant to Section 8.1(e) of this Agreement, and to carry on its business in the ordinary course substantially as carried on by NATN prior to the Effective Time, and (iii) take such other actions as may be necessary or appropriate to enable the Surviving Corporation to meet the Revenue Goal necessary to enable the Members to earn the full Contingent Payment. Without limiting the generality of the foregoing:
- Corporation as may be necessary to enable it to achieve the goals set forth in the Business Plan for the Surviving Corporation attached as an exhibit to the Management Agreement to be entered into pursuant to Section 8.1(e) of this Agreement (the "Business Plan"). Without limiting the generality of the foregoing, IntraTel shall provide to the Surviving Corporation such funds (herein referred to as "Customer Acquisition Expense Advances") as may be necessary to enable the Surviving Corporation to pay \$275,000 per week in customer acquisition expenses in the event that the Surviving Corporation does not generate sufficient revenues internally (including through the utilization of borrowings from credit facilities currently in place or replacements therefore) to pay such customer acquisition expenses.
- (b) Neither IntraTel nor the Surviving Corporation shall, without the consent of a Majority of the Members, take any action to cause the Surviving Corporation to (i) sell, lease, transfer, or dispose of any of its properties, assets or cash, (ii) merge with, liquidate into or otherwise combine with any other Person, or (iii) declare, set aside or pay any dividend, or make any other distribution (in either case in the form of cash, property or assets) with respect to the stock of the Surviving Corporation.
- 7.11 Election of James K. Noble, Jr. as Director of IntraTel. No later than the fifth day prior to the Closing, James K. Noble, Jr. shall notify IntraTel if he desires to be elected as a director of IntraTel, or if he desires to have another Member eerve in such capacity. If Mr. Noble so notifies IntraTel, IntraTel shall take all necessary action to elect Mr. Noble or such nominee to such position for a term to commence on the Effective Time and to terminate upon expiration of the Management Period.
- 7.12 Funds to Pav Member Loans. On or prior to the Closing, IntraTel shall provide to SubCorp the funds necessary for SubCorp to pay (a) to James K. Noble, Jr. the outstanding principal balance, plus accrued but unpaid interest, owed by NATN to Mr. Noble as of the date of the Closing pursuant to that certain Demand Promissory

Note of NATN dated as of December 31, 1996 in the principal amount of \$711,000.00 (such promissory note being hereinafter referred to as the "Noble Promissory Note"); and (b) to Christopher G. Hodges the outstanding principal balance, plus accrued but unpaid interest, owed by NATN to Mr. Hodges as of the date of the Closing pursuant to that certain Demand Promissory Note of NATN dated as of December 31, 1996 in the principal amount of \$474,000.00 (such promissory note being hereinafter referred to as the "Hodges Promissory Note").

- 7.13 Purchase of Call Center Companies. No later than the 30th day prior to the Closing, IntraTel shall notify each of James K. Noble, Jr. and Christopher G. Hodges if it desires to purchase all of the issued and outstanding shares of NATN Services, Inc. ("NATN Services"), a South Carolina corporation that owns a call center in Charleston, South Carolina, and all of the issued and outstanding shares of North American Communications Systems, Inc. ("Communications Systems"), a South Carolina corporation that owns a call center in Spartanburg, South Carolina. If IntraTel so notifies Messrs. Noble and Hodges, IntraTel and Messrs. Noble and Hodges shall enter into at Closing the Stc. Purchase Agreement referred to in Section 8.1(g) of this Agreement.
- 7.14 Actions to Avoid, and Notices of Breaches of Representations and Warranties. NATN shall: (a) take such actions so that NATN's representations and warranties in this Agreement remain true and correct and shall not take any action that would cause such representations and warranties to cease to be true and correct; and (b) inform IntraTel promptly of any facts or circumstances that could be reasonably expected to constitute or result in a breach of any of NATN's representations and warranties in this Agreement.
- 7.15 Additional Agreements. Subject to the terms and conditions herein provided, each of the parties hereto agrees to use its reasonable commercial efforts to take, or cause to be taken, all actions and to do, or cause to be done, all things necessary, proper or advisable under applicable laws and regulations to consummate and make effective, as soon as reasonably practicable, the transactions contemplated by this Agreement.

ARTICLE 8

CONDITIONS TO THE MERGER

- 8.1 Conditions to Merger Relating to All Parties. Consummation of the Merger is subject to the fulfillment to the reasonable satisfaction of each of the Members, NATN and IntraTel, prior to or at the Closing, of each of the following conditions:
- (a) FCC and State Public Utility Commission Approvals. Etc. All required approvals shall have been obtained from the FCC and any state public utility commission or public service commission having jurisdiction over the transactions contemplated hereby.
- (b) <u>Hart-Scott-Rodino</u>. The applicable waiting period under the HSR Act shall have expired or been terminated without either (i) a stop order, a request or other disapproval of the Merger having been issued, made or threatened by DOJ or FTC, or (b) either DOJ or FTC having requested any divestiture.
- (c) Other Consents, Authorizations, etc. All other consents, authorizations, waivers, orders and approvals of, and filings and registrations with, any governmental commission, board or other regulatory body or any nongovernmental third party which are required for or in connection with the execution and delivery by NATN of this Agreement and the consummation by NATN of the Merger or the other transactions contemplated hereby or which NATN must obtain to permit IntraTel to control the Surviving Corporation following the Merger or to permit the Surviving Corporation to continue to enjoy, in all material respects following the Merger, all of the rights and benefits of NATN, including the ability of the Surviving Corporation to continue conducting its business in all material respects in the same manner as it is now being consumed, shall have been obtained or made.
- (d) No Injunction. etc. At the Closing there shall be no judgment, decree, injunction, ruling or order of any court, governmental department, commission, agency or instrumentality outstanding against IntraTel, SubCorp or NATN which prohibits, restricts or delays consummation of the Merger or any of IntraTel's conditions to the consummation of the Merger.
- (e) <u>Management Agreement</u>. The Surviving Corporation and the Members (or an entity owned or controlled by the Members) shall have entered into a Management Agreement in substantially the form attached hereto as <u>Exhibit 8.1(e)</u>, providing for the management of the Surviving Corporation following the Effective Time.

- (f) <u>Telemarketing Service Agreement</u>. The Surviving Corporation and NSDI shall have entered into a Telemarketing Service Agreement in substantially the form attached hereto as Exhibit 8.1(f).
- K. Noble, Jr. and Christopher G. Hodges in accordance with Section 7.13 of this Agreement that it desires to purchase all of the issued and outstanding shares of NATN Services and all of the issued and outstanding shares of Communications Systems, the Surviving Corporation and each of James K. Noble, Jr. and Christopher G. Hodges shall have entered into a Stock Purchase Agreement in form, scope and substance reasonably satisfactory to each of IntraTel, Mr. Noble and Mr. Hodges, pursuant to which Messrs. Noble and Hodges shall sell to the Surviving Corporation (i) for \$1.00, all of the issued and outstanding shales of NATN Services, and (ii) for \$1.00, all of the issued and outstanding shares of Communications Systems.
- 8.2 Conditions to Merger Relating to IntraTel and SubCorp. Consummation of the Merger is subject to the fulfillment to the reasonable satisfaction of IntraTel and SubCorp, prior to or at the Closing, of each of the following conditions:
- (a) Representations and Warranties: Compliance with Covenants and Obligations. The representations and warranties of NATN contained in this Agreement shall have been true and correct at the date hereof and shall also be true and correct at and as of the Closing, except for changes contemplated in this Agreement, with the same force and effect as if made at and as of the Closing, except as such representations and warranties by their terms relate only to periods of time prior to the Closing; NATN shall have performed or complied with all agreements and covenants required by this Agreement to be performed or complied with by it at or prior to the Closing; and IntraTel shall have received a certificate of the President of NATN certifying, to the best of his knowledge, to all of the foregoing effects.
- (b) <u>Private Placement</u>. IntraTel shall have completed a private placement of convertible debt securities or debt securities and warrants of IntraTel on terms reasonably acceptable to IntraTel to enable IntraTel and SubCorp to consummate the transactions contemplated hereby.
- (c) <u>Member Releases</u>. Each of James K. Noble, Jr. and Christopher G. Hodges shall have executed and delivered to SubCorp appropriate documents evidencing payment in full of the Noble Promissory Note and the Hodges Promissory Note.
- (d) <u>NATN Membership Interests</u>. Each Member shall have delivered to SubCorp certificates or other instruments, if any, representing the NATN Membership Interests owned by such Member, duly endorsed for transfer, or such other documents necessary to transfer all of the NATN Membership Interests to SubCorp.

- (e) <u>Opinion of NATN's Counsel</u>. IntraTel shall have received an opinion of counsel to NATN, dated as of the Closing, in form, scope and substance reasonably satisfactory to IntraTel and its counsel.
- (f) <u>Additional Documents, etc.</u> NATN shall have furnished to IntraTel such additional documents and instruments as IntraTel may have reasonably requested as to any of the conditions set forth in this Section 8.2.
- 8.3 Conditions to the Merger Relating to NATN. Consummation of the Merger is subject to the fulfillment to the reasonable satisfaction of NATN, prior to or at the Closing, of each of the following conditions:
- Obligations. The representations and warranties: Compliance with Covenants and Obligations. The representations and warranties of IntraTel and SubCorp contained in this Agreement shall have been true and correct at the date hereof and shall also be true and correct in all material respects at and as of the Closing, except for changes contemplated in this Agreement, with the same force and effect as if made at and as of the Closing, except as such representations and warranties by their terms relate only to periods of time prior to the Closing; each of IntraTel and SubCorp shall have performed or complied in all material respects with all agreements and covenants required by this Agreement to be performed or complied with by it at or prior to the Closing; and NATN shall have received one or more certificates of the President of each of IntraTel and of SubCorp certifying, to the best of his or her knowledge, to the foregoing effect.
- (b) Election of James K. Noble, Jr. as Director of the Surviving Corporation. IntraTel, as the sole shareholder of the Surviving Corporation, shall have elected James K. Noble, Jr. as a director of the Surviving Corporation, for a term to commence on the Effective Time and to terminate upon expiration of the Management Period.
- James K. Noble, Jr. notifies IntraTel in accordance with Section 7.11 of this Agreement that he desires to be elected as a director of IntraTel or he desires to have another Member serve in such capacity, IntraTel shall have elected Mr. Noble or such nominee as a director of IntraTel, for a term to commence on the Effective Time and to terminate upon expiration of the Management Period.
- (d) Payment of Promissory Notes. SubCorp shall have paid to (a) James K. Noble, Jr. the then outstanding principal balance, plus accrued but unpaid interest, owed by NATN to Mr. Noble pursuant to the Noble Promissory Note, and (b) Christopher G. Hodges the then outstanding principal balance, plus accrued but unpaid interest, owed by NATN to Mr. Hodges pursuant to the Hodges Promissory Note.

- (e) Release of Guarantees. Each Member who shall have guaranteed any liability or indebtedness of NATN shall have been released therefrom by the holders of any such guarantees of such liability or indebtedness. Further, in the event IntraTel notifies James K. Noble, Jr. and Christopher G. Hodges in accordance with Section 7.13 of this Agreement that it desires to purchase all of the issued and outstanding shares of NATN Services and all of the issued and outstanding shares of Communications Systems, each Member who shall have guaranteed any liability or indebtedness of NATN Services or Communications Systems shall have been released therefrom by the holders of any such guarantees of such liability or indebtedness.
- (f) <u>Letter of Credit</u>. IntraTel shall have delivered to the Members a letter of credit, in form, scope and substance reasonably satisfactory to the Members, in the face amount of \$5,000,000.00, with the Members as the beneficiary thereof, and issued by a financial institution reproduction analysis acceptable to the Members, which letter of credit shall secure payment of the Contingent Payment to the Members.
- (g) Opinion of IntraTel's Counsel. NATN and the Members shall have received an opinion of counsel to IntraTel and Subcorp, dated as of the Closing, in form, scope and substance reasonably satisfactory to NATN and the Members and their counsel.
- (h) Additional Documents, etc. IntraTel and SubCorp shall have furnished to NATN and the Members such additional documents and instruments as NATN and the Members may have reasonably requested as to any of the conditions set forth in this Section 8.3.
- 8.4 <u>Waiver of Conditions</u>. Any party may, at its option, waive any or all of the conditions herein contained to which its obligations hereunder are subject, except that the conditions contained in Sections 8.1(a) and 8.1(b) may not be so waived.

ARTICLE 9

CLOSING

9.1 Closing. Evidence of the fulfillment or waiver of the conditions set forth in Article 8 shall be provided by the parties hereto to each other at a closing (the "Closing") to be held at the offices of Holland & Knight LLP, Suite 2000, One Atlantic Center, 1201 West Peachtree Street, Atlanta, Georgia 30309, at 10:00 a.m. local time, on the third business day following the later of (a) the closing of IntraTel's private placement as set forth in Section 8.2(b), (b) satisfaction or waiver of the conditions set forth in Sections 8.1(a) and 8.1(b), or (c) such other date as the parties may mutually agree.

- 9.2 <u>Documents to be Delivered at the Closing by NATN</u>. At the Closing, NATN shall deliver, or cause to be delivered, to IntraTel the following:
 - (a) The Articles of Merger, executed by NATN.
- (b) The Management Agreement referred to in Section 8.1(e), executed by each of the Members (or an entity owned or controlled by the Members).
- (c) The Telemarketing Service Agreement referred to in Section 8.1(f), executed by NSDI.
- (d) If applicable, the Stock Purchase Agreement referred to in Section 8.1(g), executed by Messrs. Noble and Hodges.
 - (e) The certificate referred to in Section 8.2(a).
 - (f) The releases referred to in Section 8.2(c).
- (g) The NATN Membership Interests or other documents referred to in Section 8.2(d).
 - (h) The opinion referred to in Section 8.2(e).
- (i) A copy of the resolutions of the Managers of NATN authorizing the execution, delivery and performance of this Agreement by NATN, a copy of the resolutions of the Members of NATN approving this Agreement, and a certificate of the Secretary or Assistant Secretary of NATN, dated as of the Effective Time, that such resolutions were duly adopted and are in full force and effect.
- (j) The consents or other authorizations, waivers, orders and approvals required to be obtained by NATN pursuant to Section 7.4.
- (k) Such additional certificates, opinions and other documents as IntraTel may reasonably request as to any of the conditions set forth in Section 8.2.
- 9.3 <u>Documents to be Delivered at the Closing by IntraTel and SubCorp.</u> At the Closing, IntraTel and SubCorp shall deliver, or cause to be delivered, to NATN, the following:
 - (a) The Articles of Merger, executed by SubCorp.
- (b) The Management Agreement referred to in Section 8.1(c), executed by the Surviving Corporation.

- (c) The Telemarketing Service Agreement referred to in Section 8 1(f), executed by the Surviving Corporation.
- (d) If applicable, the Stock Purchase Agreement referred to in Section 8.1(g), executed by the Surviving Corporation.
 - (e) The certificate referred to in Section 8.3(a).
- (f) Evidence of the election of James K. Noble, Jr. as a director of the Surviving Corporation referred to in Section 8.3(b).
- (g) If applicable, evidence of the election of James K. Noble, Jr. as a director of IntraTel referred to in Section 8.3(c).
- (h) Evidence of the payment of the Noble Promissory Note and the Hodges Promissory Note referred to in Section 8.3(d).
- (i) Evidence of the release of Member guarantees referred to in Section 8.3(e).
 - (j) The Letter of Credit referred to in Section 8.3(f).
 - (k) The opinion referred to in Section 8.3(g).
- (l) Copies of resolutions of the Boards of Directors of IntraTel and SubCorp authorizing the execution, delivery and performance of this Agreement by IntraTel and SubCorp and a copy of the resolutions adopted by IntraTel as sole shareholder of SubCorp approving this Agreement, and certificates of the Secretary or Assistant Secretary of IntraTel and of SubCorp, as applicable, dated as of the Effective Time, that such resolutions were duly adopted and are in full force and effect.
- (m) The consents or other authorizations, waivers, orders and approvals required to be obtained by IntraTel and SubCorp pursuant to Section 7.4.
- (n) Such additional certificates, opinions and other documents as NATN may reasonably request as to any of the conditions set forth in Section 8.3.

ARTICLE 10

TERMINATION AND ABANDONMENT

- 10.1 <u>Termination and Abandonment</u>. This Agreement and the Merger may be terminated and abandoned at any time prior to the filing of the Articles of Merger with the Secretary of State of Delaware and the Secretary of State of Georgia (the "Merger Filings"):
- (a) <u>By Mutual Action</u>. By mutual action of the Boards of Directors of IntraTel and SubCorp and by the Managers and Members of NATN
- (b) <u>By NATN</u>. By NATN if the \$250,000 Extension Fee is not received by NATN on or prior to the date of the execution of this Agreement.
- (c) By IntraTel or NATN. By IntraTel and SubCorp or by NATN if any action, suit or proceeding shall have been instituted by any person, or, to the knowledge of IntraTel (or SubCorp) or NATN, shall have been threatened by any public authority, which seeks to prohibit, restrict or delay consummation of the Merger or any of the conditions to consummation of the Merger or to the right of IntraTel to control the business of the Surviving Corporation after the Effective Time, or to subject IntraTel, SubCorp, NATN, or their respective directors, managers or officers to liability on the ground that it or they have breached any law or regulation or otherwise acted improperly in relation to the transactions contemplated by this Agreement, other than an action, suit or proceeding instituted by a person other than a public authority which, in the opinion of counsel to IntraTel and counsel to NATN, does not have a substantial likelihood of success.
- (d) <u>Deadline Date</u>. In the event that the Merger is not consummated pursuant to this Agreement on or before June 1, 1998, this Agreement may be terminated and abandoned by IntraTel and SubCorp or by NATN unless the Boards of Directors of IntraTel and SubCorp and the Managers and Members of NATN shall have agreed upon an extension of time in which to consummate the Merger.
- 10.2 <u>Procedure for Termination</u>. The termination of this Agreement by IntraTel and SubCorp or by NATN, shall be effective upon the giving of notice of such termination, stating the grounds for such termination and signed by each terminating party, to the other party.
- 10.3 Effect of Termination. In the event of the termination and abandonment of this Agreement and the Merger (a) by mutual action of IntraTel, SubCorp and NATN pursuant to Section 10.1(a); or (b) by either NATN or IntraTel and SubCorp pursuant to Sections 10.1(b), (c) or (d), no party shall have any liability hereunder (except pursuant to Sections 7.5 ("Investigation; Confidentiality"), 7.6 ("Expenses"), 7.7(b) ("No

Solicitation of Transactions; Extension Fee"), and 12.5 ("No Brokers"), which shall survive any such termination) unless such failure to consummate or fulfill a condition is within the reasonable control of either NATN or IntraTel and SubCorp, in which case such party having reasonable control shall continue to be liable hereunder.

ARTICLE 11

SURVIVAL OF REPRESENTATIONS AND WARRANTIES: INDEMNITIES

11.1 <u>Survival</u>. The representations, warranties, covenants and agreements made by IntraTel, SubCorp and the Members in this Agreement and in the schedules to this Agreement shall survive the Merger and shall not be affected by any investigation or finding made by the parties hereto prior to the date hereof or the Effective Time. The representations, warranties, covenants and agreements made by NATN shall not be affected by any investigation or finding made by the parties hereto prior to the date hereof or the Effective Time, but shall not survive the Merger; provided, however, that such non-survival with respect to NATN shall not in any way limit the liability of the Members pursuant to this Agreement.

11.2 Indemnification. Subject to the provisions of Sections 11.3 through 11.7:

- severally, agree to indemnify, defend and hold IntraTel and SubCorp harmless from and against any liability, loss, obligation, damage, cost or expense, including reasonable attorneys' fees and expenses (each auch liability, loss, obligation, damage, cost or expense referred to in this subsection (a) being referred to herein as an "IntraTel Loss") incurred or suffered as a result of (i) any material misrepresentation, material breach of warranty, or material failure to comply with any covenant or agreement, given or made by NATN or the Members in this Agreement, or (ii) the failure of the Manager (as defined in the Management Agreement) to comply with any of its obligations contained in the Management Agreement; provided, however, that if the Closing shall not occur, any sums payable pursuant to this subsection (a) shall be payable by NATN only and not by the Members; and provided, further, that any liability of a Member arising solely from a misrepresentation or breach of warranty, covenant or agreement in Article 5 hereof shall be several rather than joint, as provided in subsection (b) below.
- (b) By the Members Individually. Each of the Members, severally, agree to indemnify, defend and hold IntraTel and SubCorp harmless from and against any liability, loss, obligation, damage, cost or expense, including reasonable attorneys' fees and expenses (each such liability, loss, obligation, damage, cost or expense referred to in this subsection (b) being referred to herein as an "IntraTel Member-Specific Loss")

incurred or suffered as a result of any material misrepresentation, material breach of warranty, or material failure to comply with any covenant or agreement, given or made by such Member in Article 5 of this Agreement.

(c) By IntraTel. IntraTel hereby agrees to indemnify, defend and hold harmless NATN and the Members from and against any liability, loss, obligation, damage, cost or expense, including reasonable attorneys' fees and expenses (each such liability, loss, obligation, damage, cost or expense referred to in this subsection (c) being referred to herein as an "NATN Loss"), incurred or suffered as a result of any material misrepresentation, material breach of warranty, or material fail ro to comply with any covenant or agreement, given or made by IntraTel or SubCorput this Agreement.

(d) <u>Certain Definitions</u>. For purposes of this Article 10:

- (i) In the case of an IntraTel Loss, the "Indemnified Party" shall be IntraTel and SubCorp, and the "Indemnifying Party" shall be NATN prior to the Effective Time and thereafter shall be the Members:
- (ii) In the case of an IntraTel Member-Specific Loss, the "Indemnified Party" shall be IntraTel and SubCorp, and the "Indemnifying Party" shall be the Member against whom the claim is asserted.
- (iii) In the case of an NATN Loss, the "Indemnified Party" shall be NATN and the Members prior to the Effective Time and thereafter shall be the Members, and the "Indemnifying Party" shall be IntraTel; and
- (iv) A "Loss" shall mean an IntraTel Loss, an IntraTel Member-Specific Loss, or an NATN Loss, as the context requires.
- 11.3 <u>Limitations</u>. The indemnification obligations contained in Section 11.2 are subject to the following limitations:
- Indemnified Party under Section 11.2 unless the party claiming indemnification gives a Claims Notice on or before the second anniversary of the Effective Time, except for (i) matters covered by Section 4.11 ("Tax Matters"), as to which a Claims Notice may be given at any time prior to the date upon which the liability to which any such matter relates is barred by the applicable statute of limitations, and (ii) those covenants and agreements in Article 2 and in Section 7.10 ("Certain Obligations of IntraTel and the Surviving Corporation following the Effective Time"), which shall expire upon the later of (A) the expiration of the Management Period, or (B) such second anniversary of the Effective Time.

- (b) NATN and the Members shall not be r quired to indemnify the Indemnified Party with respect to an IntraTel Loss unless the amount of such IntraTel Loss, when aggregated with all other such IntraTel Losses, shall exceed \$50,000 (the "Minimum Aggregate Liability Amount"), at which time IntraTel Losses may be asserted for the Minimum Aggregate Liability Amount and any amounts in excess thereof; provided, however, that the foregoing Minimum Aggregate Liability Amount shall not apply to any IntraTel Loss which results from or arises out of fraud and intentional misrepresentation on the part of NATN or the Members in this Agreement.
- (c) The maximum liability of any Member for all Losses hereunder shall be limited as follows:
- (i) For any Losses arising from (I) a misrepresentation or breach of warranty in Section 4.9 ("Governmental Authorization and Compliance with Laws"), or Section 4.14 ("Material Contracts"), or (II) a violation of Section 2(e) of the Management Agreement ("slamming," etc.), each Member shall be liable for (A) an amount equal to twelve percent (12%) of the Merger Consideration actually received by such Member as of the date of such Loss if the Claims Notice with respect to such Loss is given on or prior to the first anniversary of the Effective Time, and (B) an amount equal to six percent (6%) of the Merger Consideration actually received by such Member as of the date of such Loss if the Claims Notice with respect to such Loss is given after the first anniversary of the Effective Time but on or prior to the second anniversary of the Effective Time.
- (ii) For any other Loss, each Member shall be liable for such Member's pro rate share of such other Loss, determined by multiplying the amount of such Loss by the Liability Percentage set forth beside such Member's name, as shown on Schedule 11.3 to this Agreement, and in no event shall the aggregate amount of all such Losses for which any Member shall be responsible exceed the Maximum Dollar Liability set forth beside such Member's name, as shown on Schedule 11.3 to this Agreement.
- of circumstances that have resulted, or which may result, in a Loss for which an Indemnified Party intends to seek indemnification under Section 11.2, or (b) receip, by an Indemnified Party of written or oral notice of any demand, claim or circumstances which, with or without the lapse of time, the giving of notice or both, would give rise to a claim or the commencement (or threatened commencement) of any litigation that may result in a Loss (an "Asserted Liability"), such Indemnified Party shall give notice thereof (a "Claims Notice") to the Indemnifying Party. Any Claims Notice shall describe the Loss or the Asserted Liability in reasonable detail, and shall indicate the amount (estimated, if necessary) of the Loss that has been or which may be suffered by such Indemnified Party. The Claims Notice may be amended on one or more occasions with respect to the amount of the Asserted Liability or the Loss at any time prior to final

resolution of the obligation to indemnify relating to the Asserted Liability or the Loss. If a Claims Notice is not provided promptly as required by this Section, an Indemnified Party nonetheless shall be entitled to indemnification by the Indemnifying Party to the extent that the Indemnifying Party is unable to establish that it has been prejudiced by such late receipt of the Claims Notice. Notwithstanding the foregoing sentence, however, if the Claims Notice is not provided prior to compromise or payment of any Asserted Liability by the Indemnified Party, the Indemnified Party shall only be entitled to indemnification by the Indemnifying Party to the extent that the Indemnified Party is able to establish that the Indemnifying Party has not been prejudiced by such late receipt of the Claims Notice.

- 11.5 Opportunity to Contest. The Indemnifying Party may elect to compromise or contest, at its own expense and with counsel of its choice reasonably acceptable to the Indemnified Party, any Asserted Liability. If the Indemnifying Party elects to compromise or contest such Asserted Liability, it shall, within 30 days (or sooner, if the nature of the Asserted Liability so requires), notify the Indemnified Party of its intent to do so by sending a notice to the Indemnified Party (the "Contest Notice"), and the Indemnified Party shall cooperate, at the expense of the Indemnifying Party, in the compromise or contest of such Asserted Liability. If the Indemnifying Party elects not to compromise or contest the Asserted Liability, fails to notify the Indemnified Party of its election as herein provided or contests its obligation to indemnify under this Agreement, the Indemnified Party (upon further notice to the Indemnifying Party) shall have the right to pay, compromise or contest such Asserted Liability on behalf of and for the account and risk of the Indemnifying Party. Anything in this Section to the contrary notwithstanding, (i) the Indemnified Party shall have the right, at its own cost and for its own account, to compromise or contest any Asserted Liability, and (ii) the Indemnifying Party shall not, without the Indemnified Party's written consent, settle or compromise any Asserted Liability or consent to entry of any judgment which does not include an unconditional term releasing the Indemnified Party from all liability in respect of such Asserted Liability. In any event, the Indemnified Party and the Indemnifying Party may participate, at their own expense, in the contest of such Asserted Liability. The Indemnifying Party and the Indemnified Party shall cooperate fully with each other as to all Asserted Liabilities, shall make available to each other as reasonably requested all information, records, and documents relating to all Asserted Liabilities and shall preserve all such information, records, and documents until the termination of any Asserted Liability. The Indemnifying Party and the Indemnified Party also shall make available to each other, as reasonably requested, its personnel, agents, and other representatives who are responsible for preparing or maintaining information, records, or other documents, or who may have particular knowledge with respect to any Asserted Liability.
- 11.6 <u>Subrogation Rights</u>. In the event that the Indemnifying Party shall be obligated to indemnify the Indemnified Party pursuant to this Article 11, the Indemnifying Party shall upon payment of such indemnity in full, be subrogated to all

rights of the Indemnified Party with respect to the Loss to which such indemnification relates; <u>provided</u>, <u>however</u>, that the Indemnifying Party shall only be subrogated to the extent of any amount paid by it pursuant to this Article 11 in connection with such Loss.

11.7 Indemnification Payments. Subject to the terms hereof and unless contested pursuant to Section 11.5, an Indemnifying Party shall pay to the Indemnified Party the full amount of any Loss (other than a Loss resulting from an Asserted Liability) under this Article 11 within ten days of receipt of the Claims Notice thereof and the full amount of any Loss resulting from an Asserted Liability within ten days of the date such contest is terminated or the date upon which a final non-appealable judgment or award is rendered.

ARTICLE 12

MISCELLANEOUS

- 12.1 Knowledge. "Knowledge" with respect to any individual, shall mean such information either actually known by such individual or which such individual reasonably should have known. With respect to any entity, "knowledge" shall refer to such information as was actually known by any officer, director or manager thereof.
- 12.2 Remedies. Absent fraud, misconduct or bad faith, the remedies contained in Article 2 and Article 11 of this Agreement shall be the exclusive remedies for any breach or violation of the agreements, covenants, obligations, representations or warranties set forth in this Agreement; provided, however, that such provisions shall not prevent a party from seeking equitable remedies (including specific performance and injunctive relief) in connection with any such breach or violation.
- 12.3 <u>Waiver</u>. The failure of any party hereto at any time or times to require performance of any provisions hereof shall in no manner affect the right to enforce the same. No waiver by any party of any condition, or the breach of any term, provision, warranty, representation, agreement or covenant contained in this Agreement or the other agreements contemplated hereby, whether by conduct or otherwise, in any one or more instances shall be deemed or construed as a further or continuing waiver of any such condition or breach or a waiver of any other condition or of the breach of any other term, provision, warranty, representation, agreement or covenant herein or therein contained.
- 12.4 Notices. Any notices or other communications required or permitted hereunder shall be sufficiently given if either (a) delivered personally or by courier, (b) transmitted by telecopy mechanism, provided that any notice so given is also sent for delivery as provided in clause (a) or mailed as provided in clause (c), or (c) sent by

registered or certified mail, postage prepaid, addressed as follows (or to such other address or person as any party shall have designated by notice to the other party):

If to IntraTel or SubCorp. or to

Mr. Murray Swanson 28050 US 19 North

NATN after the Effective Time: Clearwater, Florida 34621 Telecopier: (813) 221-7951

with copies to:

Benjamin Bronston, Esq.

Nowalsky, Bronston & Gothard 3500 N. Causeway Boulevard

Suite 1442

Metarie, Louisiana 70002 Telecopier: (504) 831-0892

Emanuel S. Cherney, Esq.

Kaye, Scholer, Fierman, Hays & Handler, LLP

425 Park Avenue

New York, New York 10022-3598

Telecopier: (212) 836-8689

If to NATN

North American Telephone Network, L.L.C.

before the Effective

Suite 550

Time:

Atlanta, Georgia 30319 Attn: James K. Noble, Jr.

4151 Ashford Dunwoody Road

Telecopier (888) 788-0008

with a copy to:

Charles D. Ganz, Esq.

Holland & Knight LLP

Suite 2000

One Atlantic Center

1201 West Peachtree Street, N.E. Atlanta, Georgia 30309-3400 Telecopier: (404) 881-0470

If to NSDI:

National Service Direct, Inc. 4151 Ashford Dunwoody Road

Suite 550

Atlanta, Georgia 30319

Attn: Christopher G. Hodges Telecopier: (888) 788-0008 with a copy to:

Charles D. Ganz, Esq. Holland & Knight LLP

Suite 2000

One Atlantic Center

1201 West Peachtree Street, N.E. Atlanta, Georgia 30309-3400 Telecopier: (404) 881-0470

If to the Members: Mr. James K. Noble, Jr.

455 Shadowlawn Road Marietta, Georgia 30067 Telecopier: (770) 984-2166

Mr. Christopher G. Hodges 3475 Oak Valley Road, N.E.

Unit #2920

Atlanta, Georgia 30326

Telecopier: (404) 869-0190 (call before sending by telecopy)

Mr. Hans Kasper 480 Wilde Green Drive Roswell, Georgia 30075

Telecopier: (770) 754-3130 (call before sending by telecopy)

Mr. Ronald C. Levitt 350 Dunhill Way Court Alpharetta, Georgia 30005 Telecopier: (770) 753-9486

- 12.5 No Brokers. No broker's or finder's fee or commission is due or payable from or by any of the parties hereto, nor has any such fee or commission been earned by any other third party on behalf of any of the foregoing in connection with the negotiation and execution of this Agreement or in any other manner affecting or involving the businesses of NATN or in connection with the negotiation or execution of this Agreement, or the consummation of any transaction contemplated hereby. NATN agrees to indemnify and save IntraTel and SubCorp harmless from and against any and all claims or demands for broker's or finder's fees or commissions by or from any person or persons whatsoever, based on any arrangement made by NATN. IntraTel agrees to indemnify and save NATN harmless from and against any and all claims or demands for broker's or finder's fees or commissions from any person or persons whatsoever based on any arrangement made by IntraTel or SubCorp.
- 12.6 Counterparts; Facsimile Delivery This Agreement may be executed in two or more counterparts, each of which shall be deemed an original, but all of which

together shall constitute one and the same instrument. Any party may deliver an executed copy of this Agreement and an executed copy of any documents contemplated hereby by facsimile transmission to another party except when the law expressly requires physical delivery with respect to certificates representing membership interests or other special types of documents, and such delivery shall have the same force and effect as any other delivery of a manually signed copy of this Agreement or such other document.

- 12.7 <u>Headings</u>. The headings herein are for convenience of reference only, do not constitute a part of this Agreement, and shall not be deemed to limit or affect any of the provisions hereof.
- 12.8 <u>Variation and Amendment</u> This Agreement may be varied or amended at any time by the Boards of Directors of IntraTel and SubCorp, by the Managers and Members of NATN and by the Board of Directors of NSDI, by written instrument executed by the parties hereto prior to filing the Articles of Merger with the Secretary of State of Delaware and the Secretary of State of Georgia.
- 12.9 <u>Schedules</u>. Any matter described or included in any schedule delivered herewith in response to any disclosure obligation hereunder shall be deemed disclosed for all other purposes of this Agreement. The cross references contained in the schedules delivered herewith to particular provisions of this Agreement are included therein for convenience only and shall not be deemed a part of the schedules delivered herewith or affect the construction thereof.
- 12.10 Severability. If any term or other provision of this Agreement is invalid, illegal or incapable of heing enforced by any rule of law or public policy, all other terms and provisions of this Agreement will nevertheless remain in full force and effect so long as the economic or legal substance of the transactions contemplated hereby is not affected in any manner adverse to any party hereto. Upon any such determination that any term or other provision is invalid, illegal or incapable of being enforced, the parties hereto will negotiate in good faith to modify this Agreement so as to effect the original intent of the parties as closely as possible in an acceptable manner to the end that the transactions contemplated by this Agreement are consummated to the extent possible.
- 12.11 <u>Miscellaneous</u>. This Agreement (a) constitutes the entire agreement and supersedes all prior agreements and understandings, both written and oral, among the parties, with respect to the subject matter hereof; (b) is not intended to confer upon any other person any rights or remedies hereunder; (c) shall not be assigned, by operation of law or otherwise; <u>provided</u>, <u>however</u>, that any Member may assign his rights hereunder or all or part of his NATN Membership Interests, but may not delegate his duties or obligations hereunder; and (d) shall be governed in all respects, including validity, interpretation and effect, by the laws of the State of Georgia.

FROM : STARTEL!

FROM : Ron Levitt

PHONE ND. : 5844562959

Feb. 27 1998 18:27PM P2

Sent by: MURRAY SNAHSON . Reserved - - 8787/96 - 7148941

17088690648; 02/27/98 7:47PN; Jeeffer #595; Page 2/3

PHONE NO. 1 778 753 9406

Feb. 27 1998 88:56PH P2

IN WITHERS WITERISCH, the parties becate here exceed this Agreement to be seemed on the date first above written.

InteresTele

[DOMPORATE SEAL]

PITRATEL GROUP, LTD.

Bub Corps

[CORPORATE SEAL]

DITRATEL ACQUISITION/NATH CORPORATION

enjamin W. Bronston

FROM : STARTEL:

PHONE ND. : 5044562959

Feb. 27 1998 10:27PM P3

Sent by: MURRAY SHANSON

"Recelved: "B/E7/85 TreePut

FROM : Non Levitt

17088890848; 02/27/88 7:48P4; Juffer #505; Pege 3/3

PAGNE NO. : 778 763 9466

Fab. 27 1898 88:5871 P3

LOCSPORATE SEAL!

NORTH AMERICAN TRLUPRONS METWORK, LLG.

EXHIBIT B

UNITED STATES ECURITIES AND EXCHANGE COMMISSION D.C. 20549

FORM 10-QSB QUARTERLY REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934

For the quarterly period ended Sept 30, 1997

OR	
TRANSITION REPORT PURSUANT TO SECTION OF THE SECURITIES EXCHANGE ACT OF 19	
For the transition period from	to
Commission file number	r 33-053963
INTRATEL GROUP,	LTD.
(Exact Name of Small Business Issuer	as Specified in Its Charter)
Delaware	72-1265159
State or Other Jurisdiction of Incorporation or Organization	IRS Employer Identification Number
28050 US Hwy 19 N - Suite 202, C	
(Address of principal executive offi	
(813) -797-90	
Registrant's Telephone Number,	
Not Applicat	
(Former Name, Former Address, and For Since Last Rep	mal Fiscal Year, if Changed
Check whether the issuer(1) filed all rep Section 13 or 15(d) of the Securities Exc 12 months (or for such shorter perio the file such reports), and (2) has been sub- for the past 90 days.	hange Act of 1934 during the past it the registrant was required to
Yes X No	
The number of shares outstanding of each common shares, as of the late	
Class Preferred Stock, \$.0001 par value	Outstanding at Sept. 30, 199
Common Stock, \$.0001	1,527,620
her serne	Outstanding Securities
1	

<PAGE>

INDEX TO FORM 10-QSB

INDEX

				•				PAGE
PART	I.	FINAL	CIAL INFORMAT	ION				
	Item	1.	Financial Sta	tements				
	Conde	nsed	Financial Star	tements:				
			Balance Sheet: r 31, 1996	s as of Sept	30, 199	7 and		1
			Statements of Sept 30, 1997		for the	Three Months	3	2
	Conde	nsed	Statement of	Stockholder:	' Equity			3
			Statements of Sept 30, 1997		for the	Three Months		4
:	Notes	to (ondensed Final	ncial States	ments			5
:	Item	2.	Management's I Financial (sis of s of Operati	ons	5
PART	II.	OTHER	INFORMATION					
:	Item	1.	Legal Proceed:	ings				6
:	Item	2.	Changes in Sec	curities				6
:	Item	3.	Defaults Upon	Senior Secu	rities			6
:	Item	4.	Submission of	Matters to	a Vote o	f Security-H	folders	6
:	Item	5.	Other Informat	tion				6
;	Item	6.	Exhibits and F	Reports on E	orm 8-K			6
SIGNA	TURES							7
				0				
< PAGE :	>							
				TRATEL GROUP DENSED BALAN (Unaudite	CE SHEET			
						SEPT 30, 1997	19	BER 31, 96
				ASSETS				
CURREN Cas	-	SETS:				5 134,635	\$ 1	88,365

Investments Accounts Receivable Prepaid Expenses	350	1,180 308,633 109
rapasa sopulate		109
Total Current Assets	491,900	498,287
FIXED ASSETS:		
Equipment and Furniture	86,703	64,529
Accumulated Depreciation	(37,681)	(27, 565)
Net Cost	49,022	36,964
OTHER ASSETS:		
Deposits	45,083	
Organization Costs	100,000	
Due From Infinet Software, Inc.	2,182,866	10,000
TOTAL ASSETS	\$2,868,879	\$ 545,251
141VII VIII114	72,000,075	
LIABILITIES AND STOCKHOLDE	RS' EQUITY	
CURRENT LIABILITIES:		
Accounts payable	\$ 342,708	\$ 11,376
Accrued Expenses	212,762	199,491
Note Payable, Shareholder	1,174,155	15,000
Current Deferred Tax Liability	25,800	22,500
Total Current Liabilities	1,755,425	248,367
Defended have lambalance	2 200	
Deferred tax liability Long Term Liability	7,300 4,118,514	10,000
mond term memoratel		
Total liabilities	5,881,239	258,367
COMMITMENTS AND CONTINGENCIES		
STOCKHOLDERS' EQUITY:		
Preferred stock, \$.0001 par value;	0	0
25,000,000 shares authorized;	·	ŭ
none issued		
Common stock, \$.0001 par value;	153	103
100,000,000 shares authorized;		
1,527,620 and outstanding	/1 200 0001	
Treasury Stock Additional paid-in capital	(1,200,000) 2 05,94 3	205,943
Retained Earnings	(2,018,456)	80,788
•		
Total stockholders' equity	(3,012,360)	288,884
TOTAL LIABILITIES AND EQUITY	\$2,868,879	\$ 541,251

See accompanying notes to condensed consolidated financial statements.

4	

	1997	1 9 96
••		
Revenues:		
Commissions Long Distance	367,154 171,210	685,155
Fees	553	_
Total Revenues	538,917	685,155
Cost of Sales:		
Commission expense	233,843	400,000
Reseller and other costs	152,676	_
Total cost of sales	386,519	_
Gross profit	152,398	205,155
•		
Operating Expenses:		
Selling and marketing	11,180	11,000
General and administrative	998,472	186, 921
Total operating expenses	1,009,652	197,921
Income from operations	(857, 254)	87,234
•		
Other Income:		
Interest Income	2,034	
Total other income	2,034	-
Income before income taxes	(855, 220)	87,234
Income tax provision	0	0
Net income	(855,220)	87,234
Net Income per share	(0.00)	0.06

Average common shares outstanding	1,527,620	1,527,620

The accompanying notes are an integral part of these condensed consol dated financial statements.

<TABLE>

	COMMON	STOCK	ADDITIONAL	
,			PAID-IN	RETAINED
	SHARES	AMOUNT	CAPITAL	EARNINGS
<\$>	<c></c>	<c></c>	<c></c>	<c></c>
Balance, December 31, 1996	1,527,620	\$ 153	\$(1,163,237)	\$ (912,268)
Net Income for the nine				
Months Ended Sept 30, 1997	0	0	0	(2,100,092)
Balance, Sept 30, 1997	1,527,620	\$ 153	(1,163,237)	
a fmant na	**********			*****

 | | | |The accompanying notes are an integral part of these condensed consolidated financial statements.

3

<PAGE>

Intratel Group, Ltd. Condensed Consolidated Statements of Cash Flows for the three months ended Sept 30, 1997 and 1996

	1997	1996
Cash flows from operating activities:		
Net income (loss)	(855, 220)	87,234
Adjustment to re uncile net income		
to net cash provided by (used in)		
operating activities:		
Depreciation	3,705	542
Deferred tax liability	-	-
Decrease (increase) in accounts		
receivable	(92,924)	-
Decrease (increase) in prepaid expanses	1,185	-
Decrease in accounts payable and		
accrued liabilities	119,466	(47,000)
(Decrease, increase in income		
tax payable	-	~
Net cash (used in) provided by		
operating activities	(823,787)	134,776

.

Cash flows from investing activities:		
Due from Infinet and other	(1,335,082)	0
Net cash provided from		
investing activities	(1,335,082)	0
Cash flows from financing activities:		
Increase in notes payable - shareholders	74,155	0
Other Financing Activities	2,037,500	122,071
Net cash from (used) in financing		
activities	2,111,655	0
Net (decrease) increase in cash	(47,215)	256,847
	~	
Cash, beginning of year	181,850	395
Cash, end of year	5 134,635	S 234.708

The accompanying notes are in integral part of these condensed consolidated financial statements.

4

<PAGE>

Intratel Group, Ltd.
Notes to Condensed Consolidated Financial Statements
For the Three Month Periods Ended Sept 30, 1997 and 1996

Basis of Presentation: In the opinion of the Cor

In the opinion of the Company, the accompanying unaudited financial statements contain all adjustments, consisting only of normal recurring accruals, necessary to present fairly the Company's financial position, results of operations and cash flows for the periods presented. The results of operations for the interim periods presented are not necessarily indicative of the results to be expected for the full year.

The condensed financial statements should be read in conjunction with the financial statements and the related disclosures contained in the Company's Form 10-KSB dated May 14, 1997, filed with the Securities and Exchange Commission.

Subsequent Event: None

Item 2. Management's Discussion and Analysis of Plan of Operation

The following discussion and analysis should be read in conjunction with the Financial Statements and Notes thereto appearing elsewhere in this

-- ----- . .

The following discussion contains certain forward -looks, statements, within the meaning of the "safe-harbor" provisions of the Private Securities Litigation Reform Act of 1995, the attainment of which involve various risks and uncertainties. Forward-looking statements may be identified by the use of forward-looking terminology such as "may", "will", "expect", "believe", "estimate", "anticipate", "continue", or similar terms, variations of those terms or the negative of those terms. The Company's actual results may differ materially from those described in these forward-looking statements due to, among other factors, competition in each of the Company's product areas, dependence on suppliers, the Company's limited manufacturing experience and the evolving nature of the Company's fiber optic technology.

Results of Operations

For the quarter ended Sept 30, 1997 compared to the quarter ended Sept 30, 1996

Revenue from operations for the quarter ended Sept 30, 1997 ("1997") was approximately \$538,917, compared to approximately \$685,155 for the quarter ended Sept 30, 1996 ("1996"). This represents a 21% decrease or approximately \$146,238 from 1996. This decrease in revenues was mainly attributable to conversion of accounts from marketing agreements to accounts being directly billed by Intratel.

The gross margin remained constant at approximately 32.9% and 38.6% in 1997 and 1996, respectively. Management believes the gross margin may continue to decrease due to conversion of business currently on marketing agreements to being directly billed by Intratel.

Selling and Marketing Expenses were approximately \$11,000 for the third quarter of 1997 as compared to approximately \$11,000 for the third quarter of 1996.

General and administrative expenses were approximately \$1,009,700 for 1997 compared to approximately \$187,000 in 1996, an increase of approximately 539%. Increases in general and administrative expenses were mainly due to the merger between Intratel and Intelicom International Corporation. General and administrative costs are expected to remain high as planned acquisitions in 1997 and 1998 continue to take place.

The Company had net other income of approximately \$2034 for 1997 with approximately \$0 for 1996. The 1997 amounts were favorably impacted due to additional interest earned on higher cash balances maintained during 1997 in interest bearing accounts.

Net income (loss) for the third quarter 1997 was \$(855,220) as compared to \$87,234 for the third quarter of 1996. The decrease in net income was attributable to additional costs associated with upcoming acquisitions and increased administrative costs associated with a major change in the independent contractor program and the associated marketing effort to launch it.

Liquidity and Capital Resources

Intelicom's cash position at Sept 30, 1997 was approximately \$134,635 as compared to \$188,000 at December 31, 1996. The Company had net working capital of approximately \$491,908 at \$30, 1997.

- ------

The Company believes that available cash, together with funds generated from operations and it ingoing funding process, will be ufficient to finance the Company's working capital requirements as were as planned capital additions.

5

<PAGE>

PART II

Item 1. Legal Proceedings
Not Applicable

Item 2. Changes in Securities
Not Applicable

Item 3. Defaults Upon Senior Securities
Not Applicable

Item 4. Submission of Matters to a Vote of Security Holders
Not Applicable

Item 5. Other Information
Not Applicable

Item 6. Exhibits and Reports on Form 8-K

(a) None

(b) No reports on Form 8-K were filed during the three months ended Sept 30, 1997.

6

<PAGE>

In accordance with the requirements of the Securities Exchange Act of 1934, the registrant caused this report to be signed on its behalf by the undersigned, thereunder duly authorized.

INTRATEL GROUP, LTD.

By: /s/ CHARLES R. BRINK . Date: October 8, 1997

Charles R. Brink President

By: /s/ DAVID SPEZZA . Date: October 8, 1997

-2

- ------

EXHIBIT C

NORTH AMERICAN TELEPHONE NETWORK, LLC

DRAFT

STATEMENTS OF OPERATIONS
FOR THE YEAR ENDED DECEMBER 31, 1996
AND FROM OCTOBER 27, 1995 (DATE OF INCEPTION)
THROUGH DECEMBER 31, 1995
FOR DISCUSSION PURPOSES

	1996	1995
REVENUES Customers Related Companies	\$ 4,234,059 	\$ 0 9.248
TOTAL REVENUES	2,409 co chest 55	9,248
COST OF SALES	3.031.877	0
GROSS PROFIT	2,429,715	9,248
OPERATING EXPENSES Selling, General and Administrative Expenses Customer Acquisition Expense (Note 3) Depreciation and Amortization - Other	1,368,076 3,070,910 25,164	21,793 0 966
OPERATING LOSS	(2.034.435)	(13.511)
OTHER INCOME (EXPENSE) Other Expense Interest Expense	(9,558) (108,777)	(93)
NET LOSS	\$(2.152.770)	\$ (13.604)

NORTH AMERICAN TELEPHONE NETWORK, LLC

BALANCE SHEETS DECEMBER 31, 1996 AND DECEMBER 31, 1995



PRELIMINARY & TENTATIVE FOR DISCUSSION PURPOSES

ASSETS

A Children		
	1996	1995
CURRENT ASSETS Cash Accounts Receivable Customers, part (Note 2)	\$ 11,907 	\$ 15,726 0
TOTAL CURRENT ASSETS	1,687,071	15,726
PROPERTY AND EQUIPMENT, ast (Note 4)	176,279	0
OTHER ASSETS Due from Related Companies (Note 7) Billed Unbilled Intangibles, net (Note 5) Other	366,313 141,946 60,196 3.792	0 0 56,926 0
TOTAL ASSETS	\$ 2,435,597	<u>\$ 72.712</u>
LIABILITIES AND MEMBERS C. CURRENT LIABILITIES	APITAL DEFICIT	
Advance, from Service Provider (Note 2) Capital Lease Obligation - Current Portion Accounts Psyable (Note 6) Accrued Liabilities Due to Related Companies (Note 7)	\$ 994,546 18,316 1,333,630 393,066 643,209	\$ 0 0 0 5,316
TOTAL CURRENT LIABILITIES	3,382,767	85,316
LONG-TERM LIABILITIES Capital Lease Obligation, net (Note 10) Notes Psyable Members (Note 8)	33,204 1,185,000 4,600,971	80,000 5,316
COMMITMENTS (Notes 9 & 10)	********	5,510
MEMBERS' CAPITAL DEFICIT	(2.165,374)	(12.604)
TOTAL LIABILITIES AND CAPITAL	\$ 2,435,597	\$ 72.712

NORTH AMERICAN TELEPHONE NETWORK, LLC

DRAFT

STATEMENTS OF CASH FLOWS FOR THE YEAR ENDED DECEMBER 31, 1996 AND FROM OCTOBER 27, 1995 (DATE OF INCEPTION) THROUGH DECEMBER 31, 1995

PRELIMINARY & TENTATIVE

		FOK 015 2022
	1996	1995
CASH FLOWS FROM OPERATING ACTIVITIES Net Loss	\$(2.152.770)	\$_(13.604)
Adjustments to reconcile Net Loss to Net Cash used in Operating Activities: Depreciation and Amortization - Other	25,164	966
(Increase) decrease in: Accounts Receivable Other Assets	(1,675,164) (527,865)	(57,95 2)
Increase (decrease) in: Accounts Payable Accrued Liabilities	1,333,630 1,030,959	5,316
Total Adjustments	186.724	(51.670)
Net Cash Used in Operating Activities	(1.966.046)	(65.274)
CASH FLOWS FROM INVESTING ACTIVITIES Acquisition of Property and Equipment	(188.839)	0
Net Cash Used in Investing Activities	(188,839)	0
CASH FLOWS FROM FINANCING ACTIVITIES Net Borrowings from Service Providers Increase in Capital Lease Obligation Payments on Capital Lease Capital contributed by Members Borrowings from Members	994,546 60,000 (8,480) 0 1.105.000	0 0 0 1,000 80,000
Net Cash Provided by Financing Activities	2.151.066	81,000
NET INCREASE (DECREASE) IN CASH	(3,819)	15,726
CASH, beginning of period	15.726	0
CASH, end of period	\$ 11.907	\$ 15.726
SUPPLEMENTAL DISCLOSURES: Interest Paid	\$ 13,926	s 0
Assets acquired through Capital Leases	\$ 60,000	\$ 0

NATN-Audit F/S (97) **Balance Sheet** December 31, 1997

DRAFT

ASSETS

Current Assets		
Cash in Bank - Checking	\$	<3,661.16>
Refund Account		2,704.74
Rebate Account		957.49
Accounts Receivable		8,418,731.27
Allowance for Bad Debt		<483,343.00>
Earned Income Credit Rec.		75.74
Advance to GNT		1,275.00
Advance to NSDI		14,000.00
Advance to Target		11,500.00
Advance to IMS of Atlanta		32,000.00
	1.04	

7,994,240.08 **Total Current Assets**

Property and Equipment Furniture & Fixtures 34,432.78 186,269.73 Machinery & Equipment 175,909.99 Software <87,185.52> Accum Depreciation

309,426.98 Total Property and Equipment

Other Assets 73,765.92 Organization Cost **Accum Amort-Organization Cost** <14,048.96> 3,792.00 Deposits

Total Other Assets 63,508.96

Total Assets 8,367,176.02

LIABILITIES AND CAPITAL

Current Liabilities			
Accounts Payable	5	2,959,366.94	
Accrued Liabilities		2,977,609.99	
Credits & Adjustment Reserve		75,000.00	
Checkslasued in Excess of Cash		21,642.00	
Loan From Member- J.K. Noble Jr		895,620.00	
Loan From Member- C. Hodges		597,080.00	
Promissory Note - Frank Jones		205,000.00	
Advance From OAN		4,064,643.02	
Advance from NSDI		31,578.00	
Federal FICA & W/H Payable		9,476.13	
State WH Payable	2,894.07		
Current Portion of LT Debt		41,025.00	
	14		

Total Current Liabilities 11,880,935 15

Long-Term Liabilities Deferred Gain on Asset Sale 22,042.00 Long Term Debt 62,226.00

Unaudited - For Management Purposes Only

NATN-Audit F/S (97) Balance Sheet December 31, 1997

ASSETS

DRAFT

Current Assets	
Cash in Bank - Checking	\$ <3,661.16>
Refund Account	2,704.74
Rebate Account	957.49
Accounts Receivable	8,418,731.27
Allowance for Bad Debt	<483,343.00>
Earned Income Credit Rec.	75.74
Advance to GNT	1,275.00
Advance to NSDI	14,000.00
Advance to Target	11,500.00
Advance to IMS of Atlanta	32,000.00

Total Current Assets 7,994,240.08

 Property and Equipment
 34,432.78

 Furniture & Fixtures
 34,22.78

 Machinery & Equipment
 186,269.73

 Software
 175,909.99

 Accum Depreciation
 <87,185.52>

Total Property and Equipment 309,426.98

Other Assets
Organization Cost 73,765.92
Accum Amort- Organization Cost <14,048.96>
Deposits 3,792.00

Total Other Assets 63,508.96

Total Assets \$ 8,367,176.02

LIABILITIES AND CAPITAL

Current Liabilities		
Accounts Payable	5	2,959,366.94
Accrued Liabilities		2,977,609.99
Credits & Adjustment Reserve		75,000.00
ChecksIssued in Excess of Cash		21,642.00
Loan From Member- J.K. Noble Jr		895,620.00
Loan From Member- C. Hodges		597,080.00
Promissory Note - Frank Jones		205,000.00
Advance From OAN		4,064,643.02
Advance from NSDI		31,578.00
Federal FICA & W/H Payable		9,476.13
State WH Payable		2,894.07
Current Portion of LT Debt		41,025.00
	-	CONTRACTOR OF THE PERSONS

Total Current Liabilities 11,880,935.15

Long-Term Liabilities
Deferred Gain on Asset Sale 22,042.00
Long Term Debt 62,226.00

Unaudited - For Management Purposes Only

NATN-Audit F/S (97) Balance Sheet December 31, 1997

DRAFT

Total Long-Term Liabilities

84,268.00

Total Liabilities

11,965,203.15

Capital

Members' Contributions Retained Earnings

1,000.00 <2,273,005.98> <100,000.00>

Distribution to Shareholders Net Income

<1,226,021.15>

Total Capital

<3,598,027.13>

Total Liabilities & Capital

8,367,176.02

NATN-Audit F/S (97) Income Statement For the Twelve Months Ending December 31, 1997

Year to		
Revenues	1 742 622 61	4.00
LD Revenue - Dedicated	1,743,023.51	6.05
LD Revenue - Switched	27,031,793.06	93.90
Prepaid Phone Cards	179.00	0.00
Insp/Motiv Prepd Phone Cards Start-up Checks/Fundraiser	6,886.00	0.02
Other Revenue	300.00	0.00
Gain (Loss) on Sale of Assets	19.96 4,409.00	0.00
Calli (LOSS) oil Sale oil Asses	4,409.00	0.02
Total Revenues	28,786,610.53	100.00
Cost of Sales		
LD Provider Charges	10,938,111.59	38.00
LINE COST	83,097.23	0.29
LINE COST	1,656.53	0.01
Calling Card Platform	120,078.68	0.42
PIC Change Charges	90,864.18	0.32
The change changes	70,004.10	4.32
Total Cost of Sales	11,233,808.21	39.02
Gross Profit	17,552,802.32	60.98
Expenses		
List Expense	525.00	0.00
LD Refunds	59,081.44	0.21
LD Rebates	29,061.75	0.10
Advertising	74,572.09	0.26
Telemarketing Expense	8,294,392.27	28.81
Resellers' Expenses	13,228.29	0.05
Sales Commission	11,425.00	0.04
Bank Charge	12,955.62	0.05
Merchant Bank Charges	56,046.44	0.19
Customer Service Expenses	812,730.87	2.82
Bad Debt Expense	400,972.00	1.39
Unbillables - OAN	805,050.89	2.80
Adjustments - OAN	1,039,085.63	3.61
LEC Charges - OAN	2,383,788.81	8.28
OAN Charges	1,775,182.61	6.17
Other Assoc. Charges	<220,533.42>	<0.77>
Amortization Expense	479.16	0.00
Depreciation Expense	74,626.00	0.26
Prepaid Calling Card	1,809.34	0.01
Dues and Subscriptions	11,539.58	0.04
Equipment Lease	37,068.82	0.13
Rent - Furniture	407.46	0.00
Rent - Offices	119.295.13	0.41
NDM Startup Charges	2,930.40	0.01
Data Center Charges	43,562,70	0.15
Equal Access Recovery Charges	36,648.26	0.13
Gifts & Donations	613.98	0.00
LD Installation and Setup	2,613.29	0.01
Hardware Expense	8,784.90	0.03
Programming/Software	347,864.51	1.21
Insurance - Group Health	35,183.68	0.12
Insurance - General Liability	817.00	0.00



For Management Purposes Only

merican Telephone

NATN-Audit F/S (97) Income Statement For the Twelve Months Ending December 31, 1997



